

DISSERTAÇÃO DE MESTRADO

HUMANIDADES MÉDICAS E PSIQUIATRIA: UM ESTUDO DE CASO DAS OBRAS DE VIRGINIA WOOLF

MANUELA VIANNA BOEIRA

ORIENTADOR: FLÁVIO PEREIRA KAPCZINSKI
CO-ORIENTADOR INTERNACIONAL: KAY REDFIELD JAMISON

PORTO ALEGRE, MAIO DE 2019.



DISSERTAÇÃO DE MESTRADO

MEDICAL HUMANITIES AND PSYCHIATRY: A CASE STUDY OF THE WORKS OF VIRGINIA WOOLF

MANUELA VIANNA BOEIRA

ORIENTADOR: FLÁVIO PEREIRA KAPCZINSKI CO-ORIENTADOR INTERNACIONAL: KAY REDFIELD JAMISON

PORTO ALEGRE, MAIO DE 2019.



DISSERTAÇÃO DE MESTRADO

HUMANIDADES MÉDICAS E PSIQUIATRIA: UM ESTUDO DE CASO DAS OBRAS DE VIRGINIA WOOLF

Dissertação de mestrado apresentada ao Programa de Pós Graduação em Psiquiatria e Ciências do Comportamento como requisito para obtenção de título de Mestre.

MANUELA VIANNA BOEIRA

ORIENTADOR: FLÁVIO PEREIRA KAPCZINSKI
CO-ORIENTADOR INTERNACIONAL: KAY REDFIELD JAMISON

PORTO ALEGRE, MAIO DE 2019.

MANUELA VIANNA BOEIRA

MEDICAL HUMANITIES AND PSYCHIATRY: A CASE STUDY OF THE WORKS OF VIRGINIA WOOLF

Dissertação apresentada como requisito parcial para obtenção de título de Mestre em Psiquiatria à Universidade Federal do Rio Grande do Sul, Programa de Pós-Graduação em Psiquiatria e Ciências do Comportamento.

Porto Alegre, de de 2019.
A comissão Examinadora, abaixo assinada, aprova a Dissertação "Medical Humanities and Psychiatry: a case study of the works of Virginia Woolf", elaborada por Manuela Vianna Boeira como requisito parcial para a obtenção do grau de Mestre em Psiquiatria e Ciências do Comportamento.
Prof. Dr. Flávio Pereira Kapczinski - Orientador
Prof. Dr. Ives Cavalcante Passos - Coorientador
Dr. Marcelo Pio de Almeida Fleck – Membro da banca
Dr. Cláudio Eizirik – Membro da banca
Dr. Glen O. Gabbard – Membro da banca
Dra. Simone Hauck – Membro suplente da banca
——————————————————————————————————————

Dr. Julio Cesar Walz – Membro suplente da banca

ACKNOWLEDGEMENTS

In writing this dissertation I have acquired many debts. Some of these I must acknowledge here.

Firstly, I would like to thank the Department of Psychiatry of The School of Medicine of the Universidade Federal do Rio Grande do Sul – UFRGS and its professors. I am particularly indebted to the professors and colleagues at the Laboratory of Translational Psychiatry. I am particularly grateful to Dr. Marcia Kauer Sant'Anna, Dr. Pedro Magalhães and Dr. Adriane Rosa, with whom I have learnt a great deal over the past years.

I am especially grateful to the Johns Hopkins School of Medicine and to its Department of Psychiatry in particular. I am forever grateful to the team at Meyer 4 – Adult and Young Adult Inpatient Ward. Among these team, I owe special thanks to the following residents: Claire Zachik M.D, Melissa Shepard M.D and to my now, dear friends, Prashant Sharma D.O and Natalie Gukasyan M.D. Among Meyer 4 amazing team of doctors, I can not find words to describe my gratitude towards Dr. Fernando Sampaio Goes M.D, Dr. Raymond De Paulo Jr M.D and Dr. Karen Lee Swartz M.D. I must also thank the amazing nursing staff from Meyer 4, most specially Lory Berlim, who I am also proud to call a friend.

I am extremely grateful for the many conversations and insights from Dr. Meg. Chisolm M.D, with whom I share a passion for the Medical Humanities.

I must also thank the curator and the librarians at the Henry W. and Albert A. Berg Collection of English and American Literature within the New York Public Library. They were most welcoming and they went to great lengths so that I could see/read the largest

possible number of unpublished letters by Virginia Woolf while I was researching at the Berg Collection.

I would also like to extend my gratitude to the librarians at the Mortimer Rare Book Collection at Smith College, Northampton, MA, USA.

I also ought to thank the librarians at the E.J Pratt Library at the University of Toronto, which have granted me access to the Virginia Woolf Special Collection while I was in Canada. Last, but not least, I should thank the librarians at the University of Sussex, who bared with my enormous amount of e-mails and enquires on Virginia Woolf's diaries in their possession.

I should also thank three amazing researchers from McMaster University who have helped me gain access to various papers and articles while I was there; Dr. Bianca Wollenhaupt Aguiar, Dr. Bianca Pfaffenseller and Dr. Taiane Cardoso. I miss you girls!!!

I feel I must also thank Dr. Maria Rita Drumond Viana, a post-doctorate researcher at the University of Toronto and a UFSC professor of English Literature I befriended while researching at the University of Toronto and in whom I was fortunate enough to find a good listener and also a passionate researcher of letters and diaries of other Victorian writers.

I must also express my gratitude to Smith College Emeritus Professor R. Jackson Wilson, whom I have known since childhood and who have always encouraged me to pursue my research in literature and with whom I had some of the most illuminating debates about this dissertation.

I am deeply grateful to my co-advisor, Professor Ives Cavalcante Passos M.D PhD, for his never- ending patience and support.

Finally, I must thank Dr. Kay Redfield Jamison, for embarking on this journey with a student she barely knew but who was persistent enough to convince her to accept to come on board this uncertain journey.

Most of all, I am thankful to my advisor, Professor Flávio Pereira Kapczinski, who has embraced my ideas from day one and who has kept faith in me, when a sensible person would not have done so. I have learnt an enormous amount from him, not only in the field of Psychiatry and Neuroscience, but also on how a true scholar ought to be. It has been simply an honor to have been her student.

I must also thank my parents, Beatriz Agrifoglio Vianna and Nelson Boeira, for sharing their love for books and for literature with me, not to mention my dad's library. I must also thank my mother for introducing me to the works of Virginia Woolf by lending me *A room of one's own* at the age of 14.

FOR THE YEAR OF THE INSANE

A PRAYER

(Anne Sexton)

Closer and closer

Comes the hour of my death

as I rearrange my face, grow back,

grow undeveloped and straight-haired.

All this is death.

In the mind there is a thin alley called death

And I move through it as

through water.

My body is useless.

It lies, curled like a dog on the carpet.

It has given up.

(excerpt from the poem For The Year Of The Insane, A Prayer)

RESUMO

Essa dissertação é uma tentativa de integrar os resultados conceituais e empíricos de pesquisas contempoâneas acerca do transtorno bipolar e dos resultados do estudo da biografia da escritora Virginia Woolf e de seus materiais escritos (romances, ensaios, poemas, cartas) que ela produziu durante a sua vida. Essa dissertação examina as relações entre os fatores do transtorno bipolar de Virginia Woolf e a progressão da sua condição mental, a qual culminou com o seu suicídio. A incidência do transtorno bipolar, bem como de experiências traumáticas e do suicídio entre indíviduos criativos e reflexivos, especialmente escritores e poetas é discutida, utilizando Virginia Woolf como um exemplo e um parâmetro.

Palaras chave: Transtorno bipolar. Saúde mental. Trauma. Suicídio. Virginia Woolf.

ABSTRACT

This dissertation is an attempt to integrate conceptual and empirical contemporary

research results on bipolar disease and the results of a study on the biography of Virginia

Woolf and the written materials (novels, essays, poems, letters, etc.) she produced during her

life. It examines the links between the environmental factors of Virginia Woolf's bipolar

disease and the progression of her mental condition which ended with her suicide. The

incidence of bipolar disease, trauma experiences and suicide among creative and self-

reflective individuals, specially writers and poets, is discussed, using Virginia Woolf as an

example and a parameter.

Key words: Bipolar disorder. Mental health. Trauma. Suicide. Virginia Woolf.

SUMMARY

INTRODUCTION
MATERIALS AND METRODS
CHAPTER 1. WHAT IS "MEDICAL HUMANITIES"?
CHAPTER 2. A NOTE ON NARRATIVE AND ITS RELEVANCE FOR THE STUDY OF MENTAL DISORDERS
CHAPTER3. BIPOLAR DISORDER: BRIEF HISTORY, GENERAL OVERVIEW AND ILLNESS TRAJECTORY
CHAPTER 4. VIRGINIA WOOLF – A BRIEF OVERVIEW OF HER FAMILYAND HER GENES
CHAPTER 5. VIRGINIA WOOLF AND BIPOLAR DISORDER
CHAPTER 6. TRAUMA AND BIPOLAR DISORDER IN VIRGINIA WOOLF 38
CHAPTER 7. SUICIDE, BIPOLAR DISORDER AND THE CREATIVE PROFESSIONS41
CHAPTER 8. CREATIVITY AND BIPOLAR DISORDER
CHAPTER 9. SPECIAL ARTICLE: VIRGINIA WOOLF, NEUROPROGRESSION, AND BIPOLAR DISORDER
CHAPTER 10. VIRGINIA WOOLF'S FINAL DAYS AND SUICIDE 61
BIBLIOGRAPHY67

INTRODUCTION

In this dissertation we explore the links between the environmental factors of Virginia Woolf's bipolar disease and the progression of her mental condition which ended with her suicide. For the sake of clarity we distinguished between two kinds of environmental factors. First, her social interactions and experiences (family life, childhood abuse, education, marriage, close and intimate frienships, professional contacts, historical events, etc.) which impacted her life in various ways. Second, her cultural texts (literary writings, essays, letters, etc.) which directly or indirectly (imaginatively, metaphorically, etc.) express her personal testimonies on her and other people's experiences, emotions, moods, values, reactions and other states of mind – most specially, on her illness and related sufferings.

We may say, rather improperly, that those writings – which, in addition to her creative talents, manifest the amplitude and intensity of her self-scrutiny and self-reflection – can be read as "personal clinical reports" of her origins, stages and progression of her disease. Obviously, those "personal clinical reports", in order to be validated, should have been confronted with the medical reports of the various physicians/psychiatries and health personel who examined her. Unfortunately, we were unable to gain access to those materials, something we shall try to do in the next steps of our research.

The hypothesis that support this research movement is the idea that the creative writings and similar products of creative writers affected by mental disorders can provide confirmation of recent relevant results of contemporary medical research, draw attention to aspects of the their clinical pictures not investigated as yet, and hopefully, offer new clues for

the theoretical hypotheses and empirical studies on bipolar disease and eventually to other mental disorders.

In trying to explore and integrate many distinct elements usually studied by the human sciences and the research results produced by medicine (in this case, neuroscience) we were confronted complex problems when organizing the various strings and levels of our argument in this dissertation.

In trying to produce a satisfactory written version of this type of argument we wanted to present in this text we became aware of the methodological divide between the human sciences/literature and the medical sciences. that the human sciences/literature and the medical sciences and its relevance. Each of those intelectual endeavors have their own distinct assumptions, methodologies and models of thought, characteristics that must be respected when we try to integrate their different perspectives. For instance, the medical sciences usually resort to explanations in terms of the causes of actions and events; whereas the human sciences/literature basically explore explanations in terms of the meaning of actions and events – that is to say, they examine and consider things from the 'actor's' point of view. The proper analysis of literary texts as narratives in medical humanities requires attention to this distinction.

Unfortunetely, we lack appropriate literary devices or text models to orient us on how to write scientific documents which artfully combine those two intelectual perspectives. I am fully aware that this shortcoming led me to to develop an written argument with multiple layers, not always clearly connected.

In addition to that, on a more theoretical level, I realized that each one of the diverse dimensions of the environmental and the biological factors which conform the bipolar disease have their own specific causal or meaningful effect in the resulting clinical disorder. In the future, efforts to construct models capable of describing and analyzing the relative force of each one of those "levels of causal/meaning effect" and their reciprocal influence in the genesis and progression of bipolar disease may eventually help us to better understand the interplay of the biological and existential aspects of that mental disorder.

The argument of this dissertation is organized in ten chapters (or nine, if we consider that one of the chapters is the article we have already published). Each chapter presents the information and the analysis needed to complement the discussions which are presented in subsequent chapters. Chapter 1 offers a general view of Medical Humanities and its major lines of inquiry and its pertinence to medical education and research. Chapter 2 briefly presents the definition of narrative and suggests how narrative analysis maybe useful for the interpretation of texts of creative writers with a history of mental disorder. Chapter 3 presents the concept of bipolar disease (BD) and describes the identifying traits of BD-I and BD-II. Chapter 4 introduces Virginia Woolf's family and its psychiatric history, as well as her own. Chapter 5 verses about Virginia Woolf and Bipolar Disorder. Chapter 6 discusses the different types of traumatic events that may antecipate the age onset of the disease and may be used as predictors of poor prognosis of bipolar disorder. Chapter 7 explores the correlation between bipolar disorder and suicide. Chapter 8 presents data which show the relationship between the incidence of bipolar disorder in individuals in the creative arts, specially writers and poets. The article that we have already published at the Revista Brasileira de Psiquiatria corresponds to Chapter 9, and Chapter 10, which is the final one, corresponds to Virginia Woolf's very last days.

MATERIALS AND METHODS

Throughout this study, we have used, as our main source of research *The Letters of Virginia Woolf, volumes 1 to 6*; *The Diaries of Virginia Woolf, volumes 1 to 5* as well as the five volumes of *The Autobiography of Leonard Woolf.* We have also used some of the letters available at the Henry W. and Walter A. Berg Collection of English and American Literature within the New York Public Library. In addition to that we consulted part of the letters at the Francis Hooper Collection of Virginia Woolf available at the Mortimer Rare Book Collection at Smith College. We also have gained access to some material pertaining to the Hogarth Press, obtainable at the E. J Pratt Library of the University of Toronto.

In order to gain further understanding into Virginia Woolf's mind and creative genius, we have read some of her books such as *Orlando*, *To the Lighthouse*, *The Waves*, *a Room of Ones' Own*, *Between the Acts and on Being Ill*. We have also consulted some of the lectures she has given throughout her life.

Regarding methodology, we have tried to adjust and integrate materials from diverse sources, such as Virginia Woolf's writings of various sorts, academic books on creativity and the arts, papers on narrative analysis and, most importantly, papers on bipolar disorder and neuroprogression, as well as papers linking bipolar disorder in individuals in creative professions.

I have tried to generate some exploratory literary and autobiographical texts of Virginia Woolf and the progression of her bipolar disease in order to identify clues that can be

used as "markers" of her mental health decline, which led to her suicide. Results of this study support meaning-making hypothesis. These hypotheses will have to be validated by neuroscientific methods at a later date.

CHAPTER 1. WHAT IS "MEDICAL HUMANITIES"?

Medical Humanities is now a well – established multidisciplinary field, which studies and incorporates theories, concepts, methodologies and lines of investigation drawn from disciplines as diverse as the social sciences, philosophy, bioethics, literature, linguistics and discourse analysis, cultural studies, health public policies, as well as medicine itself and other health related courses.

Over the past thirty years or so, numerous academic programs – mainly in the northern hemisphere, Australia and New Zealand – specialized scientific journals, research network groups and conferences have either been formed or have gained additional strength and interest. The number of conferences on the field has also been growing exponentially over the past decade. University programs on Medical Humanities are currently quite well – established in the United Kingdom, the United States, Canada, Sweden, Germany and Australia and the number of programs on the field, whether on a BSc or an MA/MSc or PhD level has been growing in many other countries in Europe, Asia and the Middle East.

The focal aim of the Medical Humanities is to provide a broader view on human health and explore its many dimensions and its impact upon the entire landscape of human experience. It does not attempt (or should not attempt) to substitute for the well-established and successful empirical methods (experiments, measurements, classifications, descriptions, established medical or medicine related questionnaires used to assess a number of diseases, etc.) of contemporary medical science, but to complement them, supplying additional clues,

concepts and different viewpoints that may be useful to better understand illness and suffering.

In the specific case of mental disorders, in which a patients' self-understanding (or lack thereof) of their disease is of great significance and importance, the stories they might tell themselves and others about their predicament, their conflicts, their sufferings and pains and their emotions may be considered as an essential part of a patient's "clinical picture" – and in that sense offer important material for the identification of important details on symptoms and mental processes, suggest additional clues for further investigation and, hopefully, new therapeutic pathways/strategies and processes.

The attention to the subjective perception and self-understanding of the patient – no matter how distorted they might be – requires empathy, sensitivity and a *finesse* for capturing distinctive details on the different types of narratives they employ to express their "ups and downs" (and eventually, on their evolution) of their mental disturbances. Very often, a new stage of the emotional disorder is associated with a "new story" (or narrative) about his/her self-attributed identity and the causes of the evolution (or involution) of his emotional life. In other words, the patient adopts new meanings to interpret the elements of life experiences – the previous stages of his disorder, for instance.

Literature, has traditionally described and explored in detail and, in many cases, with great subtlety, the actions the moods, the emotions, the motivations, the morals, the sensitivities and the sense of intimacy, as well as the thoughts and dreams of human beings. Those descriptions and explorations have usually taken the form of narratives that integrate an individual's subjective dimensions to a wider picture, that of its social life. Throughout this

process, literature should also draw our attention to the impact of interpersonal relationships on an individual's or group of individual's life/lives.

Of particular interest are those cases of literary men and women who suffer from mental disorders. They exhibit both the objective personal experience of the disease, as well as the talent and skill to construct narratives. In addition, writers tend to be, in most cases, reflexive people, to whom the process of imagining and constructing narratives is a central dimension of their self-image and self-respect. Therefore, they may provide unusual and useful insights and clues on their mental disorders, to be further explored by other contemporary psychiatric lines.

Similar examples can be provided for other areas and disciplines which encompass the Medical Humanities, however, we shall focus on literature given the aim of this dissertation.

CHAPTER 2. A NOTE ON NARRATIVE AND ITS RELEVANCE FOR THE STUDY OF MENTAL DISORDERS

Narratives are commonly understood as accounts by which a story is told, be it fictional or not. In general, stories present selected subjective or objective events arranged in a more or less articulated temporal sequence or process. Ideally, the explanatory power of should articulate their exhibit narratives parts to continuity, coherence and comprehensiveness. Historical writings, novels, poems, biographies, letters, dramas, films, diaries, newspapers articles, oral reports, etc. are among the many diverse forms narratives acquire. Narratives requires a narrator (or author/authors) who presents the story (in whatever format) to an audience. More precisely, narratives suppose an author and agents/actors responsible for/involved in the actions and events reported. They also presuppose a specific social scenario/context and a corresponding language which count as conditions for their intelligibility for its public. Narratives provide meanings, contexts and perspectives which are personally significant/relevant to the characters described/involved in the stories and often to the narrator/writer himself.

In this sense, physical, mathematical, chemical, economic and similar reports are not narratives, since they abstract from the subjective point of view of the individuals whose behavior, emotions and thoughts can be described or explained by the processes and causal connections they identify. In short, they ignore the meanings that are relevant to the human agents involved. It goes without saying that this "segregation" in no way affects the scientific value of their methodologies and research results.

In its broadest sense, narratives of patients have always been present in medicine in the form of clinical reports of one sort or another. Yet, until the advent of bioethics and its emphasis on the perspective of the patients, their narratives were rarely explored in their own terms. Instead, they were mostly interpreted as signs of symptoms to be explained in terms of their biological, chemical, neurological, etc. causes and their significance. Obviously, we must distinguish the attention to and empathy with the patient which was frequent in the good clinical practice and the use of patients' narratives to explain the origin of their complaints/diseases.

Psychiatry and psychoanalysis were an exception to this mode of thinking and to this methodology. In mental disorders, the self-understanding of the patient – his perceptions and eventually his ideations about his pains, sufferings, emotions, conflicts, dilemmas, etc. – is arguably part of his/her disease. The patient describes his symptoms as a temporary of more or less permanent part of his existence, as significant dimension of his ordinary experiences and as a central component of his subjectively conceived identity, or self.

Those self-descriptions are expressed in the form of narratives, which may be more or less articulated and self-conscious. They often have a motivation bearing on the patient's choices and decisions and may even become stable dispositions. Thus, close attention to those narratives of lived experiences may contain important clues for the identification of important biological and neurological processes with causal relevance.

Thus, a case may be made in favor of attentive scrutiny of narratives of patients who overly self-absorbed and conscious of the different aspects of their mental disorder and try to articulate/integrate (or fail to) them in the entire landscape of their lives. Of particular interest

are writers who in their writings meticulously describe experiences of mental disorders and the social contexts in which they occur.

Finally, of overwhelmingly importance are writers – whose activity and talent consists in constructing narratives – and who are themselves affected by serious mental disorders and testify in one form or another (novels, diaries, letters, oral exchanges, etc.) about the details and trajectory of their predicament. The medical information they provide deserve our most careful examination and analysis.

CHAPTER 3. BIPOLAR DISORDER: BRIEF HISTORY, GENERAL OVERVIEW AND ILLNESS TRAJECTORY

Bipolar disorder is the second most prevalent psychiatric disease worldwide, being responsible for a great number of psychiatric hospitalizations worldwide and claiming the lives of many others. It is estimated that a patient with Bipolar Disorder (BD) may face, on average, four hospitalizations during a lifetime, a fact that is not only severe because it greatly affects a patient's prognosis and chances of a leading a "normal "life, but also heavily burdens countries' economies. The USA is estimated to have spent directly and indirectly, in the year 2009 alone, 30.7 and 120.3 billion dollars, respectively, while in the UK, the annual cost for the NHS (National Health Service) for the years 2009-2010 with Bipolar Disorder was estimated to be around the £342 million. It has also been proved that individuals with BD are far more likely to miss work, missing an average of 18.9 working days a year if compared employees without BD, which, on average, miss only 7.4 days of work a year.

Early descriptions of the disease date back to ancient and medieval writers as mentioned in *The Bipolar Book: History, Neurobiology, and Treatment.* Accounts from 19th century doctor Jules Baillanger from (1809–1890) from the prestigious Parisian Hôpital Pitié-Sâlpetrière first described Bipolar Disorder as *Insanity of Double Form* at a lecture at the Académie de Medicine in 1854. His colleague from the same hospital, Jean - Pierre Falret (1794–1870) also claimed to have found a similar disorder several years earlier, which he coined *Circular Insanity*, which he described as a disease with manic and depressive periods, followed by some euthymic intervals.

At the end of the 19th century, Austrian psychiatrist professor at the Universities of Heidelberg and Munich in Germany, further explained the condition his fellow French colleagues had attempted to describe. He coined the term *Manic – Depressive Insanity*, which included what we currently know as Bipolar Disorder and Major Depressive Disorder. Nowadays, physicians in general, and psychiatrists and psychologists in particular, follow the major two diagnostic classification systems available worldwide when it comes to making a diagnosis of either Bipolar Disorder type I or Bipolar Disorder type II, that is to say, they either refer to the International Classification of Diseases, 11th edition (ICD-11) or to the Diagnostic and Statistical Manual of the American Psychiatric Association (DSM -5).

Bipolar Disorder type I, or BP-I, as we shall mention it from here on, is considered to be the most severe type of Bipolar Disorder given that individuals who suffer from it not only experience episodes of severe depression, but also maniac episodes that may last from 7 (seven) days to many months, if untreated. According to Merikangas, patients diagnosed with BP-I may also present either elevated mood and three other mania-related symptoms, irritable — mood plus four other mania-related symptoms. These mania — related symptoms may range from an individual's increased activity or agitation, an exaggerated sense of well-being and self-confidence (in some cases bordering psychosis or becoming an actual psychotic episode), a decreased need of sleep, unusual talkativeness and pressured speech to distractibility and poor-decision making.

Patients who experience BP - I with mixed states or mixed features, are known to present a higher rate of suicide attempts (38% vs 9%) and report higher levels of anxiety, irritability and agitation (average composite severity score of 4.1 vs 3.4). They also present a higher level of dissatisfaction with treatment response (22% vs 14%), if compared to patients

with no depressive symptoms or depressive symptoms ranging from 0-2 (all P<0.05). Bipolar disorder with mixed episodes patients' are also more likely to suffer from substance abuse disorders and anxiety disorders. Patients with BP-I with mixed episodes generally have a poorer prognosis, according to Swann et al and Goldberg and McElroy.

Bipolar Disorder type II (BP - II), on the other hand, is characterized by periods of depression and hypomanic episodes or periods. Patients are defined as having BP- II if they have had at least one major depressive episode (MDE) and at least one hypomanic episode with four or more days in length and unmistakable change in functioning.

Unfortunately, it is known that an accurate diagnosis of BD - I or BD - II can take up to ten years to be reached. That is especially true in patients with BD - II, who tend to seek help only when depressed and tend to overlook or even "like" their hypomanic episodes, which patients often find useful for working purposes as they may feel more "alert" and require less sleep during these times. Many find it very positive because they feel they can "get more done" in less time. Others may appreciate these bouts of hypomania due to the additional sex drive it may give them. Exacerbated or increased self-confidence may also prevent patients from realizing that something is amiss.

That is unfortunate, however, as an increasingly large body of research has proven that the later one is properly diagnosed, the worst is one's prognosis. The later the prognosis, the worse is the range of adverse effects such as increased severity of depression, decreased quality of life and greater likelihood to attempt suicide.

The McLean-Harvard International First Episode Project also highlights the need to be constantly alert as patients who were hospitalized in a first psychotic episode were assessed at baseline and at 24 months. As Taylor et al have stated,

[...] of 500 participants completing the study, the authors found that initial diagnosis of bipolar I disorder remained stable in 96.5% of cases. However, over one quarter of changes in diagnosis were from another psychotic illness at baseline to bipolar I disorder after two years. It is therefore important to be alert for the development of BD in people with early psychosis.

The onset of BD is often in young adulthood. In a study published in 2010 in the Journal of Affective Disorders, Baldessarini RJ et al found that the median onset for BD – I was 24.3 years and 30.1 for B.D-II. A recent study carried out in the UK found some very interesting data relating mean age of BD – I onset with a patient's family history of mood disorders and also some markers of clinical severity such as higher rates of suicidality, more rapid-cycling, more mania and depression episodes' as well as greater scores of lifetime measures of depressive symptoms. This study found that individuals whose history checked some or all of the points above mentioned had a mean onset age of 18.9, another group had a mean onset age of 28.3, whereas the third group was found to have a mean age onset of 43.3 years of age.

As for the course of the Bipolar Disorders, it is known that the number of episodes, especially manic ones, have a detrimental effect in the course of the disease. The greater the number of episodes a patient has throughout his/her life, the poorer its chances of leading a normal life become. Neuroprogression, a term coined by both professors and researchers Michael Berk M.D and Flávio Kapczinsky M.D are unequivocal about the fact that the greater the number of episodes, the worse is a patient's likely outcome.

The term "neuroprogression" has been used to define the pathological reorganization of the central nervous system along the course of severe mental disorders. In bipolar disorder (BD), neural substrate reactivity is changed by repeated mood episodes, promoting a brain rewiring that leads to an increased vulnerability to life stress. The term "neuroprogression" has been increasingly used to define the pathological reorganization of the central nervous system (CNS) along the course of severe mental disorders. This reorganization could arise as the result of several insults, such as inflammation and oxidative stress. In bipolar disorder (BD), neural substrate reactivity is changed by repeated mood episodes, ultimately promoting a brain rewiring that leads to an increased vulnerability to life stress.

It has been proven that recurrent episodes, especially manic ones, increase a patient's vulnerability to other episodes and to treatment response. Episode-dependent deterioration patterns have been found in serum biomarkers such as brain-derived neurotrophic factor (BDNF), in neuroimaging and functioning and in an increase in oxidative stress and proinflammatory biomarkers. A deficit in neuroprotection has also been noted, as was a decrease in cognitive and functioning performance.

CHAPTER 4. VIRGINIA WOOLF – A BRIEF OVERVIEW OF HER FAMILY AND HER GENES

Writing an overview on a deceased person's life is a difficult chore in itself, writing about Adeline Virginia Woolf, who is not only deceased, but also one of the most acclaimed British writers of all times is a twice as hard as Virginia Woolf, as she was and is known by most, was and still is a source of great interest and debate, not only due to her highly significant contributions to literature, but also because of her complex personal life.

Adeline Virginia Woolf, née Stephen, was born on January 25th, 1882 in a house in High Park Gate, South Kensington, London. She was, no doubt, born into a privileged household, at least in terms of social standing and intellect. Her parents, Sir Leslie Stephen KCB*, born on November 28th, 1832, was a well-known literary critic, author, and an enthusiastic mountaineering. He was educated at Eton College, King's College London and Trinity Hall, Cambridge, where he was a *wrangler*, that is to say, a student who obtained first-class honors. He obtained a B.A in 1854 and an M.A in 1857, both from the University of Cambridge.

Virginia's mother, Julia Prinsep Stephen (née Jackson) was born on February 7th, 1846 in Calcutta, where her father was a physician. She was known for her exquisite beauty and was a celebrated Pre-Raphaelite model. Her mother came from a very well-off family and she spent her childhood and adolescence at Little Holland House, in what is now Melbury Road, Kensington, London. Her family was very close to names such as painters William Holman-

Hunt and Edward Burne-Jones, and to poet and sculptor Thomas Woolner and writer John Ruskin.

Virginia Woolf was born from both her parents second marriages. Her father had been previously married to Harriet Marian Thackeray, widely known as Minnie, daughter of British novelist William Makepeace Thackeray. Minnie and Leslie had a daughter, Laura Makepeace Stephen, about whom we shall write more later. Minnie passed away in 1875, living a desperate grieving husband behind and a daughter who had started to show symptoms that was something "odd" about her development.

Virginia's mother had become a widow in 1870, after only 3 years of marriage to Herbert Duckworth, with whom she had three children – George Duckworth, born in 1868, Stella Duckworth, born in 1869 and Gerald Duckworth, born in 1870.

Leslie Stephen and Julia Duckworth got married on March 28th, 1878. Leslie was then 46 and Julia 32. Soon after the marriage was consumed the family began to expand. They had four children, of their own. Vanessa Stephen was born in 1879, Thoby Stephen was born in 1880, Virginia was born in 1882 and Adrian Stephen was born in 1883. The children of the couple and the children of their previous marriages lived under the same roof at 22, Hyde Park Gate, London until Laura Makepeace Stephen was sent to the Earlswood Asylum for Idiots and Imbeciles in 1983, at the age of 23. She was discharged in 1897 and, although she initially spent some holidays with the family in St. Ives, she would spend the vast majority of her life under the care of professionals at a farm in Sandrige, near St Albans, where she lived with Dr. Corner and his family. Dr Corner was Laura's Stephen doctor at Brook House, the last institution in which she was a boarder, so, when Dr. Corner moved to a farm with his

family, he took Laura with him. Laura Stephen passed away on February 9th 1945, at The Priory, Roehampton. She was seventy-four.

Despite this dissertation being about Virginia Woolf, we find it is of extreme relevance mentioning her parents and her siblings, given that most of her close (and not so close) family suffered from one, or a combination of mental disorders and, in Laura's case, what is now known as autism and psychosis. There are a number of scholars who also believe she, just like Virginia, was subjected to sexual molestation from her half-brothers George and Gerald Duckworth. We have chosen to start by highlighting Laura Makepeace Stephen's case not only because she was the first child, but also because data on her is most scarce and, having been institutionalized most of her life, a brief explanation of what she may have suffered from and gone through may, we believe, shed light, on Virginia's mental disorder.

There are a number of theories on which mental disorders Laura might have had, but we are from the opinion she had autism and either comorbid psychosis or perhaps an extremely severe comorbid BD -I.

Virginia's father, Leslie Stephen was known for his unpredictable mood swings and is known to have had a troubled childhood and difficult parents. He was most likely cyclothymic, a heterogenous condition and, as such, it is distinguished from major mood disorders by its chronicity and subsyndromal quality (i.e. symptoms are less severe and fewer in number, or insufficient duration). Although they may resemble the bipolar disorders due to the presence of mood swings, cycles are short and patients do not experience frank psychotic episodes. According to Kottler, even though Leslie's unpredictable bouts of anger and depression did terrify his children, they do not fit the description of patients who suffer from

either BD-I or BD-II. Leslie's father was known to have a "double nature" and Leslie's brother, Fitzjames was also afflicted by a decaying mental health, developing dementia and ended up by dying a poor man. Both Leslie Stephen and his father are known to have been, despite their rigid exterior and of both being vigilant over their children's behavior, very sensitive men, with tendencies to sink into depression, feelings of worthlessness and hopeless dejection.

Julia Prinsep Stephen, Virginia's mother, is also known to have experienced a number of depressive episodes, which are made clear in Gillespie and Steele's book on *Julia Duckworth Stephen: Stories for Children, Essays for Adults*. This book is essential for those who may want to have a better understanding of Julia Stephen's psyche and intellect, which is often portrayed as having been intellectually shallow.

Virginia Woolf's sister Vanessa and their brother Adrian were considered cyclothymic, whereas Thoby, Virginia's first brother, is known to have had hypomanic episodes, which probably allow us to classify him as having suffered from BD-II. Given Thoby Stephen's premature death from typhoid fever we can only speculate that would be diagnosis, as we do not have enough material to substantiate our hypothesis.

Finally, Virginia Woolf (nèe Stephen), whom is at the center stage of our study, is known to have endured several severe depressions and some very manic episodes. Accounts of Virginia's mental illness can be traced back to 1895, when she was only 13 years old. In her book, *A Sketch of the Past*, Virginia Woolf describes how she first felt when she saw her mother's dead body:

I remember very clearly how even as I was taken to the bedside, I noticed that one nurse was sobbing, and a desire to laugh came over me, and I said to myself as I

have often done at moments of crisis since, "I feel nothing whatever". Then I stooped and kissed my mother's face. It was still warm.

In his biography of Virginia Woolf, Quentin Bell describes her first episode, which displays both maniac and depressive symptoms:

Virginia became painfully excitable and nervous and then intolerably depressed...She went through a period of morbid self-criticism, blamed herself for being vain and egotistical, compared herself unfavorably to Vanessa and was at the same time intensely irritable.

CHAPTER 5. VIRGINIA WOOLF AND BIPOLAR DISORDER

"All extremes of feelings are allied with madness".

Virginia Woolf, Orlando

Virginia Woolf is possibly one of the most well-known writers who suffered from bipolar disorder since an early age. As thoroughly explained in previous chapters, Virginia Woolf 's family genetics, added to fact that she suffered early childhood trauma, from the death of her mother when she was only 13 years old, to the death of her half-sister Stella, followed by the deaths of her father, for whom she nurtured very mixed-feelings but whom she missed immensely when he passed away, to the untimely death of her brother Thoby, who died of typhoid fever, to the various years in which she was sexually assaulted/molested by her half-brothers George and Gerald Duckworth made her the "perfect candidate" to a severe form of BD.

Unfortunately, that proved to be the case as the numerous accounts on Virginia's breakdowns, written by herself and by others attest. Her suicide attempts are also notorious and have been studied and scrutinized in great detail. Here, we shall try to give an account of her mental distress and suicide attempts following a chronological order. We shall list her four suicide attempts first, and then describe her depressive and manic episodes.

Virginia Woolf's first suicide attempt occurred in 1904 following her father's death. Prior to her suicide attempt she was manic. She would listen to birds singing in Greek and would imagine King Edward lurking naked in the azaleas shouting obscenities. Virginia tried to leap from a first-floor window and was then institutionalized and had to lay in bed for months. It was the first time she was subjected to the then famous "Rest Cure Therapy", which was prescribed by Dr. George Savage, who was a huge fan of his American colleague Silas Weir Mitchell treatment, which he had come up with in 1873. The therapy consisted of having the patient lay in bed all day long, eat, drink plenty of milk and not have any disturbing contacts with the outside world such as reading books, writing letters, or anything that could agitate the patient or bring unpleasant memories. Plenty of sleep was also prescribed. 'I have never spent such a wretched 8 months in my life,' she wrote to a friend when the crisis had passed.

Poirier quotes a description of the cure as extracted from Mitchell's Doctor and Patient:

In carrying out my general plan of treatment it is my habit to ask the patient to remain in bed from six weeks to two months. At first, and in some cases for four or five weeks, I do not permit the patient to sit up or sew or write or read. The only action allowed is that needed to clean the teeth. In some instances, I have not permitted the patient to turn over without aid. In such cases I arrange to have the bowels and water passed while lying down, and the patient is lifted on to a lounge at bedtime and sponged, and then lifted back again into the newly-made bed. In all cases of weakness treated by rest, I insist on the patient being fed by the nurse, and, when well enough to sit up in bed, I insist that meats shall be cut up, so as to make it easier for the patient to feed herself.

Virginia Woolf's second suicide attempt occurred in 1913, one year after she had married Leonard Woolf. Her second suicide attempt was much more severe and she almost did not make it. Virginia took 100g of Veronal (*Barbital*) and, if it weren't for Leonard Woolf

and Geoffrey Keynes (who was house, surgeon at Saint Bartholomew's Hospital at that time) she would most probably have died. She required urgent medical attention so a gastric lavage was performed and, although she was between life and death for some hours, she woke up the next morning. Virginia was then sent to a private nursing home for females in Twickenham, where, once again, seclusion and the *Rest Cure* were enforced. She was placed on a regime of weight gain; four or five pints of milk daily, as well as cutlets, liquid malt extract and beef tea.

George Savage, her psychiatrist, recommended that a patient "who went in weighing seven stone six comes out weighing 12". This advice clearly made an impact on Virginia: she repeats it almost verbatim in Mrs. Dalloway when the celebrated psychiatrist Sir William Bradshaw orders "rest in bed; rest in solitude; silence and rest; rest without books . . . so that a patient who went in weighing seven stone six comes out weighing 12".

Virginia later wrote: "Leonard made me into a comatose invalid".

Even though there is a degree of truth in Virginia's complaint, she certainly wouldn't have survived much longer if it were not for Leonard Woolf's unwavering devotion and patience to keep her as safe and as far from a severe mood swing as he could. He would make her spend the mornings in bed when he foresaw an imminent crisis, he would monitor her eating and weight, her moods and her menstrual cycles. And Virginia was certainly aware of that, as we can verify in a letter, she wrote to Jacques Raverat in 1922: "unless I weigh 9 and half stones, I hear voices and see visions and can neither write nor sleep".

She also wrote to her friend and lover, Vita Sackville-West, admitting Leonard's efforts to keep her sane, as shown below: "I should have shot myself long ago in one of these illnesses if it hadn't been for him".

Virginia did not fully recover until the end of 1915. At the end of February 1915, she "entered a state of garrulous mania, speaking ever more wildly, incoherently and incessantly, until she lapsed into gibberish and sank into a coma,", according to both Leonard Woolf and Quentin Bell. By March she was being cared for by professional nurses, and was unable to see or speak to Leonard – he writes that she was "violently hostile". At times her psychotic episodes were so severe that she required four nurses to hold her down, and there was genuine doubt over whether she would ever fully recover. She remained under professional care until November, when she finally returned to Hogarth House: "I spend my spare time in bed, but I'm allowed out in the afternoons, and thank God the last Nurse is gone."

It is not known for certain the number of suicide attempts Virginia made during 1913-1915, but both her husband accounts and many of her biographers also state that she made a couple.

What I consider to be her third suicide attempt was in fact a suicide gone wrong, as one week prior to her death she attempted to drown on the River Ouse, next to the Woolf's house in Roadmell. At that point Leonard was already quite aware that she was not doing well and had shared his concerns with their friend and physician, Octavia Wilberforce, who would come to "visit" the couple once a week and check on Virginia. According to Leonard, she went for a walk and returned soaked wet, shivering and exhausted. When enquired on what had happened, she said she had stumbled on a rock and slipped into the sea.

However, we know that was not the case as she had written to her friend Ethel Smyth on March 1st, 1941, saying: "We have no future; I'm fished out of my element and lie grasping on the ground".

On March 28th 1941, Virginia Woolf made, once again, her way to the River Ouse, this time however, she would not return. She was wearing a heavy overcoat and had made sure the stones on both pockets were heavy enough for her to drown. Virginia's cane was found on the banks of the River Ouse on that very afternoon. Three weeks later, on April 19th 1941, children found her body near the bridge in Southease.

As Shirley Panken brilliantly puts it, "the sea, a major symbol in Woolf's life and fiction, represents the rhythmic nature of existence, the inexorable cycle of building up and destroying, the nonhuman life of which she so much felt a part."

CHAPTER 6. TRAUMA AND BIPOLAR DISORDER IN VIRGINIA WOOLF

The correlation between traumatic events and bipolar disorder has been widely researched; we find it of the utmost importance to discuss it here given, not only its scientific significance, but also for understanding Virginia Woolf's biography. We shall revisit some of traumatic episodes of her life in light of various researches that establish the link between traumatic events and predisposition of an individual to suffer from either BD-I or BD-II. Traumas may act as a trigger to either the onset of bipolar disorders or to its additional episodes.

According to Nierenberg, early trauma can permanently change one's response to stress, leading to hyperarousal and hypervigilance and sustained stress. An imbalance in these patients' response to pro and anti-inflammatory cytokines may lead to sustained inflammation effects in multiple organs and can lead to premature death. That might be the cause why individuals with bipolar disorder may lose up to 25 years more of their lives when compared to individuals who do not suffer from BD. In patients who suffer from BD, Leverich S and Post found that early childhood trauma can be directly associated with a greater number of subsequent manic or depressive episodes, faster cycling patterns, a greater number of suicide attempts, and an increased number of additional psychiatric and medical disorders, including higher incidence of alcohol or substance abuse.

Serious outcomes were found in those who suffered sexual abuse – which was Virginia's Woolf case – especially if other non-sexually related traumatic events are endured, which in her case, were not insignificant. Losing her mother and her sister Stella at a very

early age, having a father who displayed constant mood swings someone whom she cared for deeply and died when she was twenty-two years old, followed by the death of her brother Thoby, whom she was very close to, can only have made things harder for her to cope with.

Hammersley *et al* also found that people who had suffered early-life trauma were more prone to have hallucinations. In their study, 45% of their subjects had suffered from hallucinations at some point in their lives or continuously suffered from them, a very relevant number considering that their cohort was of 96 subjects.

When compared with subjects that had not suffered physical or sexual abuse in either childhood or during adolescence, subjects who had undergone these types of trauma also showed a higher predisposition to schizoid, schizotypal and paranoid behavior, as well as to narcissistic, borderline, histrionic or antisocial behavior. The last three being more commonly found in patients who had a history of sexual abuse.

Leverich et al also support the thesis that early traumatic experiences further decrease the age of onset of BD and is also a strong predictor of a more severe unfolding of bipolar disorder.

According to Palmier- Claus *et al*, Etain B *et al* and Upththegrove R *et al* childhood adversity has been linked to the presence of psychosys.

The broad scope of the analysis presented in *Virginia Woolf and Trauma* offers a thorough and compelling analysis of the haunting, if sometimes repressed trauma narrative that can be found embodied throughout Woolf's texts. The truthfulness of Woolf's description

of the sexual abuse she suffered is present in the autobiographical memoir *A Sketch of the Past*, whilst simultaneously seeking to avoid a direct approach to the topic at hand. Consequently, "trauma" is understood as both personal – as in the deaths of Woolf's mother, half-sister Stella, and brother Thoby, and as collective, as we shall explain below.

Woolf's manifold bouts of illness, both physical and psychological, as well as the sexual abuse she suffered by being molested by her two half-brothers, Gerald and George Duckworth, were only, or mostly, private affairs, as opposed to the nationwide trauma of the two World Wars, which affected the whole of Britain and many other nations and their citizens.

Only Toni H. McNaron, retired professor from the University of Michigan, specifically deals with Woolf's non-fictional writing by investigating the way in which Woolf tackles her early sexual abuse within her essays in the book *Virginia Woolf and Trauma*, mentioned above.

Below, we shall briefly quote some of Virginia Woolf's pieces of writing that illustrate the points above mentioned.

CHAPTER 7. SUICIDE, BIPOLAR DISORDER AND THE CREATIVE PROFESSIONS

Suicide is a major health problem. Current global suicide mortality rates amount to 1.4% of all deaths worldwide. This number might seem small, but when confronted with data from the World Health Organization (WHO), which has put it in sheer numbers – 800.000 people commit suicide every year, one realizes that it has become a serious international issue. In the last data made available by the WHO in 2016, suicide was rated as the 18th most prevalent cause of death worldwide. To put it even more plainly, one person commits suicide every forty-second. To make matters even worse, the WHO has proven that suicide is the leading cause of death amongst the 14-29 years old population worldwide.

Although suicide and suicide attempts certainly are not, by any rate limited to individuals with bipolar disorders, it has been widely researched that those who suffer from either BD-I or BD-II are more at risk of committing or attempting suicide than individuals who suffer from other mental disorders such as schizophrenia, obsessive compulsive disorder (OCD), generalized anxiety disorder (GAD), amongst others.

According to Novick *et al.*, the number of suicides and suicide attempts (SA) do not differ significantly between patients with BD-I or BD-II. However, their literature review and meta-analysis show that patients with BD-II are more prone to use violent and lethal methods than those diagnosed with BD-I. Researchers estimate that between 25% and 60% of individuals with bipolar disorder will attempt suicide at least once in their lives and between

4% and 19% will complete suicide. Yet, Novick et al point out to the need of further research focusing on BD-II patients, as, to these date, most research on suicide and SA attempts on bipolars focused on patients with BD-I, which was believed by many researchers, until recently, to be the most severe form of the disease and thus, patients are more prone to commit suicide and SA.

Much of what we know about suicide risk in bipolar disorder is from research in which the sample is comprised largely of individuals with bipolar I disorder (BPI). In the studies that do include individuals with bipolar II disorder (BP-II), researchers rarely consider BPII outcomes separately. The prevalence of suicide attempts (SA) in BP-II, particularly in comparison to the prevalence in BP-I, is an understudied and controversial issue with mixed results. Tondo, on the one hand, found significantly higher rates of suicide attempt amongst patients with BD-II when compared with those with BD-I (21.9% versus 15.9%, respectively), whereas Coryell *et al.* found significantly higher rates among individuals with BPII than among individuals with BPII (41.2% versus 17.6%, respectively). Study design and methodology seem to be the main causes for such discrepancies so we believe further studies with similar or same study design and methodology are needed before any definite claim can be made on this matter.

Now we shall move into the topic of suicide amongst artists, writers and creative individuals in general, as this was the very case of Virginia Woolf. Even though demographics, race, gender, amongst other things, may play a role in determining which individuals are more at risk of taking their lives or attempting to do so, a study of 32 occupations with a large number of suicides carried out by Stack concluded, even after a host of demographic confounders, that artists remained at a significantly higher risk of committing

suicide if compared to other professions. Preti & Motto and Preti, *et al.* have found similar results in previous studies. According to Stack, in the best designed study, artists are found to have a risk of dying from suicide versus natural causes that is 2.12 times greater than the rest of the working population. It is worth highlighting that nine studies showed that artists constituted one of the very few occupation groups where suicide risk is independent of demographic cofounders.

Research results are even gloomier for poets and writers, which exceeded the average risk of suicide among artists or those who had chosen creative careers, such as architects, for example, who, alongside with painters, are at a lower risk than the mean.

The fact that suicide and SA are higher in poets and writers may have a number of explanations, which range from economic strain due to volatile income, pressure from printing houses, fear of critics' reaction, and a tendency for severe self-criticism are considered the most important ones.

Poets and writers are especially prone to BD-I and BD-II. Kay Jamison, a professor of psychiatry at Johns Hopkins University and the author of *Touched with Fire: Manic-Depressive Illness and the Artistic Temperament*, said writers are 10 to 20 times as likely as other people to suffer manic-depressive or depressive illnesses, which lead to suicide more often than any other mental disorders do. According to Jamison, it is not surprising that these mood disorders seem most at home in the artistic mind. "The cognitive style of manic-depression overlaps with the creative temperament". Researchers have found that in a mildly manic state, subjects think more quickly, fluidly and originally. In a depressed state, subjects are self-critical and obsessive, an ideal frame of mind for revision and editing. "When we

think of creative writers, "Dr. Jamison said, 'we think of boldness, sensitivity, restlessness, discontent; this is the manic-depressive temperament".

Apart from Virginia Woolf, several well-known writers and poets such as Jack London, Cesare Pavese, Anne Sexton, Sylvia Plath, Ernest Hemingway, Vladimir Maiakowski and Primo Levi ended their lives by committing suicide.

Virginia Woolf has left us some interesting letters and diary notes on that matter, and also on how she felt about literary critics, publishing houses and the panic and dread the launching of a book caused on her. Leonard Woolf, her lifetime husband, has also written some interesting accounts on how Virginia would react on such occasions, which can be found below:

The horror is that tomorrow, after this one windy day of respite – oh the cold north wind that has blown ravaging daily since we came, but I've had no ears, or nose: only making my quick transits from house to room, often in despair – after this one's day respite, I say, I must begin at the beginning and go through 600 pages of cold proof. Why, oh why? Never again, never again (Virginia Woolf in *A Writer's Diary*).

In his diary, Leonard wrote about Virginia:

The torture began as soon as she had written the last word of the first draft of her book; it continued off and on until the last reviewer, critic, friend, or acquaintance had said his say. [...] Her major works, by increasing the strain, only increased the 'dejected rambling misery' and the grey welter'. The moment she sent back the proofs of The Waves to the printer she had to go to bed with a dangerous headache. When the book was published and before she had any criticism of it, whether from Hugh Walpole, John Lehmann, or anyone else, she wrote in her diary: I have come up here, trembling under the sense of complete failure – I mean The Waves, I mean Hugh Walpole does not like it – I mean John L. is about to say he thinks it bad. And months later she said she said she still felt her brain numb from the strain of writing The Waves. [...] We had a terrifying time with The

Years in 1936; she was much nearer a complete breakdown than she had ever since 1913.

CHAPTER 8. CREATIVITY AND BIPOLAR DISORDER

Even though the link between psychiatric diseases, mainly the bipolar disorders, but also schizophrenia (to a smaller degree) and creativity has gained momentum after the 1980's, it is fair to say that it is quite an ancient concept, so much so that Greek philosophers such as Socrates (469–399 BC) and his student Plato (428–348 BC) praised the benefits of *divine madness*, which they believed was literally a gift from the gods. According to Koutsantoni, the concept of madness, as deployed by Plato and Socrates, encompassed a wide range of states of thought and emotion, not just psychosis, but the emphasis was upon a profoundly altered state of consciousness and feeling. Aristotle's thoughts on these matters soon ensued. He focused, however, on the link between melancholia, madness and inspiration, asking the question: "Why is that all men who are outstanding in philosophy, poetry or the arts are melancholic?"

By the end of the 20th century, a number of scholars turned to this very question Aristotle once asked. The link between psychiatry and the humanities gained momentum and researchers sought to understand the symptoms and signs of mental health conditions in conjunction with the humans experiencing them, rather than separating the person from the condition. Systematic studies have demonstrated a causal nexus between creative genius and madness, specially writers and poets.

Amongst a body of research made to this day, Andreasen and Jamison are what one would call pioneers. In 1987 Nancy Andreasen studied rates of mental illness in 30 creative

writers (27 men and 3 women), 30 matched control subjects and the first-degree relatives of both groups. Andreasen's study showed very high rates of affective disorder, that is to say, 80% of the writers she interviewed had had an episode of affective illness at some time in their lives, compared to 30% found in the control group subjects'. Out of these 80% who presented some affective illness, 43% of them were bipolar disorders versus a "mere" 10% of subjects in the control group who had experienced bipolar illness.

In 1993, Kay Redfield Jamison published *Touched with Fire: Manic-Depressive Illness and the Artistic Temperament* where *she* discusses the clear link between creativity and bipolar disorder. Some years prior to the publication of this book, Jamison had interviewed 47 eminent British writers and artists and had found that, although most of the writers and artists she interviewed had been treated for depression and only 38% for bipolar disorder, many offered accounts of bursts of energy and swinging moods, which are consistent with hypomania, a trait of BD-II. Jamison concluded that there is a much higher than expected rate of bipolar disorder, depression and suicide in exceptionally creative artists and writers, as opposed to non-creative individuals. There is an important, albeit alarming statistical difference between the percentage of individuals in the general population diagnosed with manic-depressive illness – 1%, versus an astonishing 38% amongst artists and writers.

A more recent Swedish study lead by MacCabe, carried out over the course of 10 years, found a correlation between the risk of developing bipolar disorder and individuals who excelled academically, achieving A grades, specially in humanities. Another Swedish study lead by Kyaga and published in 2012 at the Journal of Psychiatric Research is worth mentioning, given the scope of its database. They performed a nested case-control study using longitudinal Swedish total population registers and, although this larger study did not find

significant evidence between individuals in the creative professions and mental disorders in general, it did find that authors suffered from schizophrenia and bipolar disorder more than twice as often as controls. Authors were also more likely to be diagnosed with unipolar depression, anxiety disorders, alcohol abuse, drug abuse, and to commit suicide. The finding of increased risk for suicide in authors might be secondary to other psychopathology, according to researchers. To investigate this further, they omitted all authors with any psychiatric diagnosis in the PNR – Personal Identity Number or "*Personnummer*" in Swedish (ICD-8: 290e315, ICD-9: 290e319, ICD-10: F00eF99). There was still a trend for authors without diagnosed psychopathology to commit suicide more frequent than controls (OR 1.45, 95% CI 0.97e2.16; p ¼ 0.07). Thus, regardless of psychopathology, being an author seemed to increase suicide risk.

Besides the above data, Kyaga et al also admitted that they had found no positive association between psychopathology and overall creative professions except for bipolar disorder.

In Virginia Woolf's case, creativity played a substantial role and it was, paradoxically, both what kept her sane and "going" and also what led her to depths of despair. To Virginia, creativity was mainly, though not only, was associated to her ability to write and read. Writing produced an exhilarating but sometimes painful effect, while writing had the ability to catapult her from gloomy moods and depressions. Her mere existence depended greatly on her being able to write and read, as we shall note in the passage below, in a letter she exchanged with Hugh Walpole on August 23rd, *1933*:

And then, being in a gloom the other night, I took down your Apple Trees (1932) and enjoyed it so much, more the second reading than the first, that I invented a theory to the effect that you being a born romantic, and I not being one,

what I like is when you turn your rich lantern upon facts, because they become rimmed and haloed with light but still remain facts in the centre. So, I want you to go on writing your memories. It struck me today on my walk that, that I like [Sir Walter] Scott's diaries better than all but three of four of his novels for this reason. And Vanessa – to end the argument – will have more fact in it than the others; what fun, if I'm there in flesh! – or my name's there.

Such a taste for a life devoted to writing may be connected with and account for her liking for history. Apart from Walpole's historic novels of herries series, Woolf liked Jules Michelet's *L'Histoire de France (1883-843)*: for example, in a letter written on 31st January 1928 to Clive Bell:

I am reading Michelet. Does it strike you that history is one of the most fantastic concoctions of the human brain? That it bears the remote likeness to the truth seems to me unthinkable." Consider the character of Louis 14th. Incredible! And those wars – unthinkable! Ought it not all be re-written instantly? Yet, he fascinates me.

In another letter from February 26th, 1928, addressed to Julian Bell, Virginia wrote: "I am reading Michelet's History of France – God knows why. I find it fascinating, but wholly fictitious. Do you think any history is even faintly true?"

What autobiographical and historical writing may have in common is facts, whether truthful or fictional. Woolf's taste for the two genres most probably helped her deal with the facts of her own life and redefine the art of the self-portrait in her letters. In a passage on the *Essays of Virginia Woolf*, she wrote: "the 'leaning-tower writers', with help from Dr. Freud, wrote more about themselves in their autobiographies 'honestly', therefore creatively".

In another entry from her fourth volume of letters Virginia wrote:

They told the unpleasant truths, not only flattering truths". That is why autobiography is so much better than their fiction or poetry. **Consider how difficult**

it is to tell the truth about oneself – the unpleasant truth; to admit that one is petty, vain, mean, frustrated, tortured, unfaithful and unsuccessful.

Woolf not only emphasizes the importance of such truthful self-analysis but also insists that such an "unconscious" state of writing should be further developed by the next generations of writers. Her creativity was based on her intense and continuous self-examination and her desire to express herself through a **highly elaborated** personal and literary writings.

Such a form of self-expression was more than simply the exercise of a natural and highly educated talent. It was her major defense against the destructiveness of her past traumas and scars they had left in her adult emotional life. In her literary life she could not avoid dealing with the themes which grew out of the intimations of her traumas and their consequences. She could not avoid confronting the ghosts which were, in part, a byproduct of her bipolar condition.

CHAPTER 9. SPECIAL ARTICLE: VIRGINIA WOOLF, NEUROPROGRESSION, AND BIPOLAR DISORDER

Manuela V. Boeira¹²

Gabriela de Á. Berni¹²

Ives C. Passos¹ ²

Márcia Kauer-Sant'Anna¹ ²

Flávio Kapczinski¹ ²

¹Programa de Transtorno Bipolar, Laboratório de Psiquiatria Molecular, Hospital de Clínicas

de Porto Alegre (HCPA), Porto Alegre, RS, Brazil

²Programa de Pós-Graduação em Psiquiatria, Departamento de Psiquiatria, Universidade

Federal do Rio Grande do Sul (UFRGS), Porto Alegre, RS, Brazil

ABSTRACT

Family history and traumatic experiences are factors linked to bipolar disorder. It is known

that the lifetime risk of bipolar disorder in relatives of a bipolar proband are 5-10% for first

degree relatives and 40-70% for monozygotic co-twins. It is also known that patients with

early childhood trauma present earlier onset of bipolar disorder, increased number of manic

episodes, and more suicide attempts. We have recently reported that childhood trauma partly

mediates the effect of family history on bipolar disorder diagnosis. In light of these findings

from the scientific literature, we reviewed the work of British writer Virginia Woolf, who

allegedly suffered from bipolar disorder. Her disorder was strongly related to her family

background. Moreover, Virginia Woolf was sexually molested by her half siblings for nine

years. Her bipolar disorder symptoms presented a pernicious course, associated with

hospitalizations, suicidal behavioral, and functional impairment. The concept of

neuroprogression has been used to explain the clinical deterioration that takes places in a subgroup of bipolar disorder patients. The examination of Virgina Woolf's biography and art can provide clinicians with important insights about the course of bipolar disorder.

Key words: Mood disorders; bipolar; suicide; stress; sexual assault; cognitive neuroscience

Family history and traumatic experiences are factors linked to bipolar disorder (BD). It is known that the lifetime risk of BD in relatives of a bipolar proband are 5-10% for first degree relatives and 40-70% for monozygotic co-twins. It is also known that patients with early childhood trauma present earlier onset of BD, increased number of manic episodes, faster cycling pattern, and more suicide attempts.

Additionally, we have recently reported that childhood trauma partly mediates the effect of family history on BD diagnosis.³ It is noteworthy that sexual abuse was associated with BD in our study, but not with major depressive disorder.³ Moreover, other authors have reported that illness trajectories are largely variable in bipolar disorder.⁴ It seems that a subset of patients may develop a neuroprogressive course associated with poor outcomes, such as suicide attempts, hospitalization, and functional and neurocognitive impairment.⁴ In light of these findings, we reviewed the biography and work of Virginia Woolf, one of the most renowned female writers of the 20th century and amongst the finest British novelists ever, and who, according to biographers, suffered from bipolar disorder.⁵,

Mental illness in Virginia Woolf's family can be traced back to James Stephen, her grandfather on her father's side. James was allegedly cyclothymic and, according to Bell, also given to self-mortification and depression. He was eventually institutionalized, after running naked through Cambridge. He died in an asylum. Virginia Woolf's parents also suffered from

mental disorders - her father was what at the time was called cyclothymic, whereas her mother suffered from depression. Sir George Savage, a prominent psychiatrist in the late 19th and early 20th centuries, diagnosed Virginia's father Leslie with "neurasthenia," a common medical term used in the late 19th to early 20th centuries. Her half-sister Laura, who spent most of her life at the Priory Hospital Southgate in London, is believed to have had some type of psychosis. Her specific mental illness (or illnesses), however, is yet unknown. Amongst Virginia's other siblings, both Vanessa and Adrian appear to have been cyclothymic, and Thoby was known to have hypomanic episodes. Hence, Virginia's BD symptoms appear to be strongly linked to her family background.

Virginia's biological inheritance translated into great risk of developing mental illness. Nevertheless, it could be argued that her disease would have been milder had she not been exposed to childhood traumatic experiences. ¹¹Virginia Woolf was sexually abused by her half siblings, George and Gerald Duckworth, for nine years. ⁶ According to De Salvo, "these experiences had spoiled her life before it had fairly begun." ¹⁰ When she was only six, Gerald molested her while the Stephen family was vacationing in St. Ives, Cornwall. George's advances would not come until seven years later, after their mother Julia had passed away. The age difference between Virginia and her brothers should be noted - Gerald was 16 years her senior, and George was 15 older. Below is Virginia's account of Gerald's first sexual move towards her. ¹²

There was a slab outside the dining room door for standing dishes upon. Once, when I was very small, Gerald Duckworth lifted me onto this, and as I sat there he began to explore my body. I can remember the feel of his hand going under my clothes; going firmly and steadily lower and lower. I remember how I hoped that he would stop; how I stiffened and

wriggled as his hand approached my private parts. But it did not stop. His hand explored my private parts too. I remember resenting, disliking it - what is the word for so dumb and mixed a feeling? It must have been strong, since I still recall it. This seems to show that a feeling about certain parts of the body; how they must not be touched; how it is wrong to allow them to be touched; must be instinctive. (Poole, 13 p. 25)

Virginia Woolf also left written accounts about how, after the death of her mother Julia, her half-brother George would enter her room and enthusiastically lie next to her and take her into his arms. She later wrote about his "violent guts of passion," and that his behavior was "a little better than a brute's." In "22 Hyde Park Gate," Virginia describes George sexual advances towards her. Virginia noticed that someone had entered her room as she was lying in bed trying to sleep. Her account is as follows:

Who? I cried.

Don't be frightened, George whispered. And don't turn on the light, oh beloved!

Beloved - and flung himself on my bed, and took me in his arms. (Poole, 13 p. 111)

De Salvo suggested that Virginia Woolf later attempted to "heal her childhood wounds" through writing. 14

Such issues may have emerged in her writing of the novel *The Wise Virgins*, which was started during her honeymoon. The novel's main character, Camilla, bears enormous resemblance to Virginia Woolf in her attitudes towards sex, such as fear and sometimes aversion to it. Even so, Virginia hoped to have children. Shortly after her wedding, Virginia was heartbroken when her doctors advised her to refrain from motherhood on account of her ongoing mental health issues. 16

As mentioned at the beginning of this article, early trauma is associated with increased number of suicide attempts in patients with BD.² Virginia Woolf's first suicide attempt happened when she was 22 years old, after her father's death.¹⁶ Having withstood her mother's and Stella's death, her father's departure triggered an even greater depression that seemed too much to bear for someone who was already in the doldrums. She tried to jump out of a window in the family's home in London. Fortunately, the window she jumped from was not high enough to cause her any major injuries. She was hospitalized for a short period but soon returned home.¹⁷ Her second suicide attempt was very serious and happened in 1913, when Virginia was 31 years old.¹⁸ This time she attempted suicide by taking 100 g of barbital. She would have died if it weren't for Leonard and two physicians, Henry Head and Geoffrey Keynes, who came to the home and pumped Virginia's stomach with a pump borrowed from St. Bartholomew's hospital. Henry Head was an English neurologist whom Virginia was supposed to see in London, and Geoffrey was the brother of an acquaintance from the Bloomsbury Group who lived in Brunswick Square. Throughout her recovery, her writing and reading were rationed, and she was only considered fully recovered in August 1914.

From 1910 to 1913 Virginia was hospitalized several times for suicide attempts, and was submitted to "rest cure therapy" at a "private nursing home for women" in Twickenham. 19 20 The therapy consisted mainly in gaining weight, sleeping, and "rest of the intellect." Virginia loathed her institutionalizations but somehow agreed that they were her only way towards recovering her sanity. Virginia went through several severe depressive and manic episodes until she committed suicide. Many of the episodes preceded the release of her books, always a cause of anxiety and self-doubt. Leonard often had her "institutionalized at home," cared by one or several nurses, depending on the severity of the episode. 21 It seems that Virginia had decided to end her life, and some scholars believe she tried to drown herself

one week before finally "succeeding at it." One evening, she arrived home soaking wet after a failed suicide attempt. According to Leonard, she looked ill and shaken but she told him that she had slipped into a dyke. 16

One week later, Virginia Woolf filled her overcoat pockets with heavy stones and headed to the River Ouse to never return. She died at the age of 59. At that time the couple was living at Hogarth House in Roadmell, East Sussex. She had been severely depressed. Despite Leonard's attempts to keep her sane (he himself was also depressed), and despite the involvement of Dr. Octavia Wilberforce's, his efforts were, sadly, of no use.

Towards the end of her life, Virginia's mental health deteriorated even further. According to her biographer's account, she became suspicious, even paranoid. He had started to doubt her publisher's praise of her soon to be published *Between the Acts*. She wanted to revise it further, but mentions in her diaries that she felt that she could no longer write, that she was "losing her art." Apparently there was some truth in her editor's lack of certainty towards her work. It is known that some of her publishers started to become ambivalent towards her work, which was a major blow to her ego. Her inability to read and concentrate, and also to perform "the simplest of tasks" such as holding a pen for long periods time was clearly unbearable to her. Throughout the last years of her life, symptoms of mood episodes typical of BD occurred increasingly more frequently, despite the fact that they were not new to Virginia. The episodes would start with sleeplessness, progressing to hearing voices.

The course of BD is highly variable, but it seems that Virginia Woolf presented a pernicious course, associated with hospitalizations, suicidal behavior, and cognitive impairment. Recently, the term neuroprogression was proposed in order to explain why a

subset of BD patients might experience a worsening of their mental health over time. ²⁴ Neuroprogression has been hypothesized as the pathological rewiring of the brain that takes place in parallel with cognitive, functional, and clinical deterioration in the course of BD. ⁴ In this sense, reductions in the volume of the fronto-limbic system and cognitive impairment have been reported as a function of previous manic episodes and hospitalizations. ²⁵ In addition, it has been proposed that trauma and number of mood episodes may show sensitization to themselves and cross-sensitization to one another, leading to residual vulnerability to further occurrences of mood episodes and faster illness progression. ²⁹ The progression of Woolf's BD seems to fit the model proposed by the hypothesis of neuroprogression - this is supported by some of her final diary entries and the suicide note she left to Leonard:

I feel certain that I am going mad again. I feel we can't go through another of these terrible times. And I shan't recover this time. I begin to hear voices and I can't concentrate. So I am doing what seems the best thing to do. (Glendinning, ³⁰ p. 323)

ACKNOWLEDGEMENTS

ICP is supported by a scholarship from Coordenação de Aperfeiçoamento de Pessoal de Nível Superior (CAPES), Brazil. FK and MK-S are research fellows from Conselho Nacional de Desenvolvimento Científico e Tecnológico (CNPq).

REFERENCES

- 1. Craddock N, Sklar P. Genetics of bipolar disorder. Lancet. 2013;381:1654-62. [Links]
- 2. Leverich GS, Post RM. Course of bipolar illness after history of childhood trauma. Lancet. 2006;367:1040-2. [Links]
- 3. Jansen K, Cardoso TA, Fries GR, Branco JC, Silva RA, Kauer-Sant'Anna M, et al. Childhood trauma, family history, and their association with mood disorders in early

- adulthood. Acta Psychiatr Scand. 2016 Jan 30., doi: 10.1111/acps.12551. [Epub ahead of print] [Links]
- 4. Passos IC, Mwangi B, Vieta E, Berk M, Kapczinski F. Areas of controversy in neuroprogression in bipolar disorder. Acta Psychiatr Scand. 2016 Apr 21., doi: 10.1111/acps.12581 [Epub ahead of print] [Links]
- 5. Caramagno TC. The flight of the Mind. Berkeley: University of California; 1992. [Links]
- 6. Bennet M. Virginia Woolf and neuropsychiatry. Netherlands: Springer; 2013. [Links]
- 7. Bell Q. Virginia Woolf: a biography. New York: Harcourt Brace Jovanovich; 1972.

 [Links]
- 8. Beard G. Neurasthenia, or nervous exhaustion. Boston Med Surg J. 1869;80:217-21. [Links]
- 9. Taylor RE. Death of neurasthenia and its psychological reincarnation: a study of neurasthenia at the National Hospital for the relief and cure of the paralysed and epileptic, Queen Square, London, 1870-1932. Br J Psychiatry. 2001;179:550-7. [Links]
- 10. De Salvo LA. Virginia Woolf: The impact of childhood sexual abuse on her life and work.

 Boston: Beacon; 1989. [<u>Links</u>]
- 11. Williams LC. Virginia Woolf's history of sexual victimization: a case study in light of current research. Psychology. 2014;5:1151-64. [Links]
- 12. Woolf V. Moments of being: a collection of autobiographical writing. 2nd ed. Orlando: Harcourt Brace; 1985. [<u>Links</u>]
- 13. Poole R. The unknown Virginia Woolf. 4th ed. Cambridge: Cambridge University; 1996.

 [Links]
- 14. De Salvo L. Writing as a way of healing: how telling our stories transforms our lives. Boston: Beacon; 2002. [Links]
- 15. Woolf L. The wise Virgins. London: Persephone; 2003. [Links]

- 16. Lee H. Virginia Woolf. New York: Vintage; 1990. [Links]
- 17. Panken S. Virginia Woolf and the lust of creation: a psychoanalytic exploration. Albany: The State of New York Press; 1987. [Links]
- 18. Franklin I. Virginia Woolf's veronal overdose, 1913 [Internet]. 2014 Dec 02 [cited 2016

 Mar 04]. wellcomehistory.wordpress.com/2014/12/02/the-near-death-of-thenovelist/ [Links]
- 19. Woolf E. The joyful, gossipy and absurd private life of Virginia Woolf [Internet].

 Newsweek. 2015 [cited 2016 Mar 04]. newsweek.com/2015/02/27/joyful-gossipy-and-absurd-private-life-virginia-woolf-306438.html [Links]
- 20. Mulvihill ME. Dancing on hot bricks Virginia Woolf in 1941 [Internet]. Rapportage. [cited 2016 Mar 04]. carlkohler.se/pdf/Mulvihill-Woolf-PDF.pdf [Links]
- 21. Szasz T. My madness saved me: the madness and the marriage of Virginia Woolf. New Jersey: Transaction; 2006. [Links]
- 22. Bell AO, Mcneillie A. The diary of Virginia Woolf. Orlando: Mariner; 1979. [Links]
- 23. Reese SJ. Recasting social values in the work of Virginia Woolf. Selinsgrove: Susquehanna University; 1996. [Links]
- 24. Berk M, Kapczinski F, Andreazza AC, Dean OM, Giorlando F, Maes M, et al. Pathways underlying neuroprogression in bipolar disorder: focus on inflammation, oxidative stress and neurotrophic factors. Neurosci Biobehav Rev. 2011;35:804-17. [Links]
- 25. Strakowski SM, DelBello MP, Zimmerman ME, Getz GE, Mills NP, Ret J, et al. Ventricular and periventricular structural volumes in first- versus multiple-episode bipolar disorder. Am J Psychiatry. 2002;159:1841-7. [Links]
- 26. Mwangi B, Wu MJ, Cao B, Passos IC, Lavagnino L, Keser Z, et al. Individualized prediction and clinical staging of bipolar disorders using neuroanatomical biomarkers. Biol Psychiatry Cogn Neurosci Neuroimaging. 2016;1:186-94. [Links]

- 27. Cao B, Passos IC, Mwangi B, Bauer IE, Zunta-Soares GB, Kapczinski F, et al. Hippocampal volume and verbal memory performance in late-stage bipolar disorder. J Psychiatr Res. 2016;73:102-7. [Links]
- 28. Rosa AR, Magalhães PVS, Czepielewski L, Sulzbach MV, Goi PD, Vieta E, et al. Clinical staging in bipolar disorder: focus on cognition and functioning. J Clin Psychiatry. 2014;75:e450-6. [Links]
- 29. Post RM, Kalivas P. Bipolar disorder and substance misuse: pathological and therapeutic implications of their comorbidity and cross-sensitisation. Br J Psychiatry. 2013;202:172-6. [Links]
- 30. Glendinning V. Leonard Woolf: a biography. Berkeley: Counterpoint; 2008. [Links] Received: March 14, 2016; Accepted: March 29, 2016

Correspondence: Flávio Kapczinski, Universidade Federal do Rio Grande do Sul (UFRGS), Av. Ramiro Barcelos, 2350, CEP 90035-903, Porto Alegre, RS, Brazil Email: flavio.kapczinski@gmail.com

Disclosure FK has received grants or research support from AstraZeneca, Eli Lilly, Janssen-Cilag, Servier, NARSAD, and the Stanley Medical Research Institute; has been a member of speakers' boards for AstraZeneca, Eli Lilly, Janssen, and Servier; and has served as a consultant for Servier. MK-S is on speaker or advisory boards for, or has received research grants, from NARSAD, the Stanley Medical Research Institute, and Eli Lilly. The other authors report no conflicts of interest.

This is an Open Access article distributed under the terms of the Creative Commons Attribution

Non-Commercial License, which permits unrestricted non-commercial use, distribution, and reproduction in any medium, provided the original work is properly cited.

CHAPTER 10. VIRGINIA WOOLF'S FINAL DAYS AND SUICIDE

"Against you I fling myself, unvanquished and unyielding, O Death!

The waves broke on the shore"

Virginia Woolf in *The Waves*

As Virginia's suicide and death approached, Leonard, always very perceptive to her mood swings and any other signs that something might have been amiss, decided to contact Dr. Octavia Wilberforce, who is known to have been Virginia Wool's last physician. Octavia practiced medicine in Montpelier Crescent, Brighton, where she lived with Elizabeth Robins. The Woolf's had met Octavia Wilberforce some years prior, when they had invited her partner to dine with them in London.

When the Woolf's were in Roadmell, usually from August to September, Octavia and Elizabeth would visit them and they would sometimes pay them a visit in Brighton. By then, Octavia Wilberforce had, according to Leonard, became Virginia's doctor to all intents and purposes.

Leonard expresses his concern in his diary:

I told Octavia and consulted her professionally. The desperate difficulty which always presented itself when Virginia began to be threatened with a breakdown – a difficulty which occurs, I think again and again in mental illness – was to decide how far it was safe to go in urging her to take steps – drastic steps – to ward off the attack. Drastic steps meant going to bed, complete rest, plenty of food and milk. But part of the disease was to deny the disease and to refuse the cure. There was always

the danger of reaching the point when, if one continued to urge her to take the necessary steps, one would increase not only the resistance but her terrible depression.

According to Leonard Woolf accounts, by February 7th, 1941 she couldn't remember why she had been depressed an wrote an entry in her diary: "Why was I depressed? I cannot remember".

On the 11th the couple travelled to Cambridge where they visited The Hogarth Press at Letchworth and saw the Principal of Newham College, Cambridge and dined at King's College, also in Cambridge, with Dadie Rylands. They were visited by Elizabeth Bowen, Vita Sackville- West and Enid Jones. Again, according to Leonard accounts, Virginia seemed to be enjoying herself a great deal.

As proof of Virginia's intense mood swings is an account written by Leonard describing a conversation that Virginia had had with Octavia Wilberforce in mid – March of 1941, when she would persuade Octavia to tell her the story of her life. Virginia was particularly fond of autobiographies and had attempted at convincing many of her friends into writing one and had written a biography on Roger Fry herself.

Leonard wrote in his diary:

Whenever Octavia came to see us, Virginia tried to get her to tell the story of her life and she had this vague idea of perhaps making it into a book. "However, there is a note from March 18th, 1941, which is distressing.

I am not sure whether early in the week she did not attempt to commit suicide. She went for a walk in the water-meadows in pouring rain and I went, as I often did, to meet her. She came back across the meadows soaking wet, looking ill and shaken. She said that she had slipped and had fallen into one of the dykes.

On Friday Octavia came to tea and I told her that I thought Virginia on the verge of danger.

By March 26th Virginia was overcome by desperate depression, her thoughts raced beyond her control; she was terrified of madness. According to Leonard: One knew that **at any moment she might kill herself.** The only chance for her was to give in and admit she was ill, but this she would not do.

At Leonard's request, and also because she was indeed quite concerned herself and did care for Virginia not only as a patient, but also as a friend, Octavia Wilberforce kept on visiting the Woolf's once a week and always brought milk and cream, which she knew were essential to Virginia, specially at such times, when she would go hours without eating or eating barely nothing. But, as Leonard stated in his autobiography, Virginia was unaware of the reason for such visits.

These visits were, so far as Virginia was concerned, just friendly visits, but I had told Octavia how serious I thought Virginia's condition was becoming and from our point of view, the visits were partly medical.

On Wednesday, March 26, I became convinced that Virginia's mental condition was **more serious than it had ever been** since those terrible days in August 1913 which led to her complete breakdown and attempt to kill herself.

Virginia Woolf final diary entry dates from March 24th, 1941, after her failed suicide attempt but four days prior to her suicide. It gives us a glimpse into what her thoughts were back then. It is quite interesting to note what she omits rather than what she openly says. Her last diary entry goes as follows:

Monday March 24

She had a <face> nose like the Duke of Wellington & great horse teeth & cold prominent eyes. When we came in, she was sitting perched on a 3 cornered chair with knitting in her hands. An arrow fastened her collar. And before 5 minutes had passed, she told us that two of her sons had been killed in the war. This, one felt, was to her credit. She taught dressmaking. Everything in the room was red brown & glossy. Sitting there I tried to coin a few compliments. But they perished in the icy sea between us. And then there was nothing. A curious sea side feeling in the air today. It reminds me of lodgings on a parade at Easter. Everyone leaning against the wind, nipped & silenced. All pulp removed. This windy corner. And Nessa is at Brighton, & I am imagining how it wd be if we could infuse souls. Octavia's story. Could I englobe it somehow? English youth in 1900. Two long letters from Shena & O. I can't tackle them, yet enjoy having them. L. is doing the rhododendrons...

As mentioned in previous chapter Virginia had attempted to kill herself one week before but was "unsuccessful". She did not admit it though but Leonard was sure that she had tried to drown on the River Ouse as he met her halfway and found her soaked wet, trembling and shaking. Her lies made Leonard even more concerned and he was thorn between having to force her into, once again, leading a "vegetative life", as he called it, something he too found inhumane and undignifying. He knew that, this time, Virginia was not willing to cooperate and get onboard the "restorative therapies" that had saved from losing her mind as well as her life before. It is interesting, and profoundly sad that, Leonard and their physician and friend, Dr. Octavia Wilberforce had decided on the 27th of March, after much consideration, that she had to be institutionalized. However, she was more determined than both of them and walked towards the River Ouse in the morning of March 28th, 1941. This time, she made sure she was wearing a heavy overcoat and that she had plenty of rocks on her pocket so that she would certainly drown. Her cane was found on the bank of the Ouse that afternoon. Her body, however, was found only three weeks later by children playing next to a bridge. Her body had been carried by the currents.

She left two poignant suicide letters (although some scholars state there she left some more), which are known by many, but, instead of first quoting the suicide letter addressed to her husband Leonard, on which she explained her motives for killing herself but mostly thanks and reassures him of everything he has done to her, we shall first quote the letter she left to Vanessa, which we find much more revealing in terms of how desperate and hopeless she was feeling at that time.

The note reads:

Sunday

Dearest, You can't think how I loved your letter. But I feel I have gone too far this time to come back again. I am certain now that I am going mad again. It is just as it was the first time, I am always hearing voices, and I shan't get over it now. All I want to say is that Leonard has been so astonishingly good, every day, always; I can't imagine that anyone could have done more for me than he has. We have been perfectly happy until these last few weeks, when this horror began. Will you assure him of this? I feel he has so much to do that he will go on, better without me, and you will help him. I can hardly think clearly anymore. If I could I would tell you what you and the children have meant to me. I think you know. I have fought against it, but I can't any longer. Virginia.

The note she left Leonard had been drafted on the 25th of March, three days prior to her suicide, which indicates that she was absolutely resolute about her decision.

Below is the letter she left to her husband Leonard:

Dearest.

I feel certain I am going mad again. I feel we can't go through another of those terrible times. And I shan't recover this time. I begin to hear voices, and I can't concentrate. So I am doing what seems the best thing to do. You have given me the greatest possible happiness. You have been in every way all that anyone could be. I don't think two people could have been happier till this terrible disease came. I can't fight any longer. I know that I am spoiling your life, that without me you could

work. And you will I know. You see I can't even write this properly. I can't read. What I want to say is I owe all the happiness of my life to you. You have been entirely patient with me and incredibly good. I want to say that – everybody knows it. If anybody could have saved me it would have been you. Everything has gone from me but the certainty of your goodness. I can't go on spoiling your life any longer.

I don't think two people could have been happier than we have been. V

Her end was anticipated in Virginia Woolf's mind and words in many occasions since her adult's life. She seems to have been waiting for a sum of an unbearable combination of subjective distresses and objective disasters that finally came in the weeks before March 28th, 1941.

BIBLIOGRAPHY

CHAPTER 1. WHAT IS "MEDICAL HUMANITIES"?

Bleakly, Alan. Medical humanities, humanities and medical education, how medical humanities can shape better doctors, Routledge, London, 2015.

Bracken, P, Thomas P, Timirni S, Asen E et al. Psychiatry beyound the current paradigma, The British Journal of Psychiatry, 201, 430-434, 2012.

Brody, H. Defining The Medical Humanities. Three concepts and three narratives, Journal of Medical Humanities. 2011, vol. 32, issue 1, 1-7.

Cole, T, Carlini, M, Carson, R, Medical humanities, na introduction., Cambridge University Press, New York, 2015.

Cole J., Gallagher S. Narrative and clinical neuroscience: can phenomenologically informed approaches and empirical work cross-fertilization?, in The Edinburgh companion to the cultural medical humanities, Edinburgh University Press, Edinburgh, 2016, 377-394.

Dolan B. Humanitas, readings in the development of medical humanities. University of California Medical Humanities Press, San Francisco, 2015.

Downie R. Medical humanities: some uses ans problems. J R Coll Physicians Edin, 2016, 288-94, 2016.

Dunbar K, Fugelsang J. Scientific thinking and reasoning, in The Cambridge handbook of thinking and reasoning, Cambridge University Press, Cambridge, 2005, 705-26.

Evan H M, Macnaughton. Should medical humanities be an multidisciplinar or an interdisciplinar study? Medical Ethics; Medical Humanities, 30, 1-4, 2004.

Holyoak K J, Morrison R G. The Cambridge handbook for thinking and reasoning, Cambridge University Press, Cambridge, 2005.

Gilman, Sander L. Illness and Image, Case studies in medical humanities, Routledge, New York, 2017.

Greenfield P. Paradigms of cultural thought, in Cambridge handbook for thinking and reasoning, Cambridge University Press, Cambridge, 2005, 663-81.

Who gets sick, who gets better, and why? Johns Hopkins Center for Medical Humanities and Social Medicine, https://www.hopkinsmedicine.org/...the johnshopkins center f...

Louis-Courvoisier, M Wenger A. How to make the most of history and literature in the teaching medical humanities: the experience of the University of Glasgow, www.medicalhumanities.com.

Macnaughton, J. Medical humanities' challange to medicine, Journal of Clinical Practice, 2011 oct 17(5) 927-932.

Macnaughton, J. The past, the present, and the future medical humanities, postmedical: a ournal of medieval cultural studies (2017), 8, 234-239.

Macnaughton, J. The humanities in medical education: contexto,, outcomes and structures, J Medical Humanities 2000; 26, 23-30.

Oyebode F. The medical humanities: literature and medicine. Clinical Medicine, 2010, vol. No. 3: 242-244.

Patel V, Arrocha J, Zhang J. Thinking and reasoning in medicine, in Cambridge handbook for thinking and reasoning, Cambridge University Press, Cambridge, 205, 727-749.

Pikkersgill M, Hogle L. Enhancement, ethics and society: towards an empirical research agenda for the medical humanities and social sciences, Med Humanit, 2011, 41: 136-142.;

Polianski I. Toward a "harder" medical humanities: moving beyond the "two cultures" dichotomy. Academic Medicine, 2012: 87: 121-127.

Shapiro J, Couleban J, Wear D, Mointello W. Medical humanities and their discontents: definitions, critiques and implications (2009). In Dolan B. Humanitas, readings in the developments of medical humanities, University of California Medical Humanities Press, San Francisco, 2015, 268-289.

Slaby J. Critical neurosciences meet medical humanities. Medical Humanities 2015 41: 16-22.

Whitehead A, Woods A (eds). The Edinburgh companion to the cultural medical humanities, Edinburgh University Press, Edinburgh, 2016.

CHAPTER 2. A NOTE ON NARRATIVE AND ITS RELEVANCE FOR THE STUDY OF MENTAL DISORDERS

Abbott H P (ed.) The Cambridge introduction to narrative, 2nd. Edition, Cambridge University Press, Cambridge, 2009.

Baldwin, C. Narrative, ethics and people with severe mental illness, Australian and New Zealand Journal of Psychiatry, 2005, 3: 1022-1029.

Berrios, G, Markova I. The self and psychiatry: a conceptual history, in Kircher T, David Anthony. The self in neuroscience and psychiatry, Cambridge University Press, Cambridge, 2003, 9-39.

Brockmeier J. Narrative scenarios: toward a culturallu thick notion of narrative, in Valsina, J ed.) Oxford handbook of culture and psychology, Oxford University Press, Oxford, 2012, 587-605;

Brown, B, Nolan P, Crawford P, Lewis A. Interaction, language, and the "narrative turn" in psychotherapy and psychiatry, Soc. Scie. Medicine, vol. 42. No. 11, 1569-1578, 1996.

Bruner, J. Life as narrative, Social Research, vol 71: n.3, fall 2004, 691-710.

Carretero M, Bermudez A. Constructing histories, in Valsina J (ed.) Oxford handbook of culture and psychology, Oxford University Press, Oxford, 2012, 625-646.

Downie R. Medical humanities: some issues and problems, Journal of the Royal College of Physicians Edinburgh, 2016: 46.

Fireman B, Flanagan O, MacVay Jr T. Narrative and consciousness: literature, psychology and the brain, Oxford University Press, Oxford, 2003.

Flanagan, O. Consciousness revisited, MIT Press, Cambridge, 1992.

Greenhalgh, T, Hurwitz. Why study narrative? BMJ 1999, Jan 2, 318 (7175) 48-50.

Herman, D (ed.) The Cambridge companion to narrative, Cambridge University Press, Cambridge, 2007.

Hermans H J M . Self-narrative as meaning con1struction: the dynamics of self-investigation, Journal of Clinical Psychology, 1999, 55; 1193-1211.

Holmes J, Narrative in psychiatry and psychotherapy, the evidence? Journal of Medical Humanities, 2000, 26; 92-96.

Jones A H. Literature and medicine, an evolving canon. https://web.archive.org/web/20160818092843.

Hydén, Lars-Christien. Illness and narrative, Sociology of Health and Illness, vol. 19, n. 1, 1997, 48-89.

Kircher T, David A. The Self in neuroscience and psychiatry, Cambridge University Press, Cambridge, 2003.

Knuuttila S. Fictionalising trauma, the aesthetic of Margerite Duras' india Circle, Helsinkki University Print, Helsinki, 2009, specially 47-48

Kokanovic R, Fiore J. Subsjectivity and illness narratives, in Subjectivity (2017) 10: 32-339.

Lewis, B. Narrative and psychiatry, Current Opinion in Psychiatry, 2011, 24: 489-494.

Lewis, B. Taking a narrative turn in psychiatry, The Lancet, Jan 04, 2014, 22-26.

McKechnie, C C. Anxieties of communication: the limits of narrative in medical humanities, Medical Humanities, published on line on May/28/2014.

McLeod J. Narrative and psychotherapy, Sage Publications, London, 1997.

Nelson, Katherine. Narrative and the emergence of a consciousness of self, in Fireman, D, Flanagan O, MacVay Jr. T, Narrative and consciousness: literature, psychology and the brain, Oxford University Press, Oxford, 2003, 17-36.

Oyebode F. Fictional narrative and psychiatry, in Advanced Psychiatry Treatment, http://apt.rcpsych.org/site/subscriptions/ Riesmann C K. Analysis of Personal Narrative, (to appear in Gubrium J F, Holstein J A. Handbook of interwiewing, Sage Publications, London, 2001, Open Read access.

Ryan M L. Toward a definition of narrative, in Herman D (ed.), The Cambridge companion to narrative, Cambridge University Press, Cambridge, 2007, 22-35.

Shapiro J. Illness narratives: reliability, authenticity and the empathic witness, Medical Humanities, Dec 2011, vol. 37, n. 2, 68-72.

Stamenov, M. Language and self-consciousness: modes of self-presentation in language structure, in Kircher T, David A. The Self in neuroscience and psychiatry, Cambridge University Press, Cambridge, 2003, 74-104.

CHAPTER 3. BIPOLAR DISORDER: BRIEF HISTORY, GENERAL OVERVIEW AND ILLNESS TRAJECTORY

Angst J, Gamma A, Benazzi F, Ajdacic V, Eich D, Rossler W. Toward a re-definition of subthreshold bipolarity: epidemiology and proposed criteria for bipolar - II, minor bipolar disorders and hypomania. Journal of Affective Disorders. 2003. Jan 73 (1-2): 133-146. (PubMed: 12507746).

Aysegul Yildiz, Pedro Ruiz, Charles Nemeroff (ed.). The Bipolar Book: History, Neurobiology, and Treatment. New York: Oxford University Press, 2015.

Bauer M, Pfenning. A Epidemiology of bipolar disorders. Epilepsia. 2005; 46 (Suppl 4); 8-13. (PubMed: 15968806).

Bortolato B, Köhler A. Cristiano, Evangelou E. *et al.* Systematic assessment of environmental factors for bipolar disorder: an umbrella review of systematic reviews and meta-analyses. Bipolar Disorders. 2017; 19: 84-96. DOI: 10.1111/bdi.12490

Goldberg J. and McElroy L S. Bipolar Mixed Episodes: Characteristics and Comorbidities. The Journal of Clinical Psychiatry. 68(10:e25 November 2007. PubMed

Grande I, Berk M, Birmaher B, Vieta E. Bipolar Disorder. Lancet (London, England). 2016, 387: 1561-1572

Kapczinski F, Vieta E, Magalães V.S Pedro and Berk, M, editors. Neuroprogression and Staging in Bipolar Disorder. 2015. 1-4, 81-92. Oxford University Press. Oxford, England, UK.

Merikangas KR, Jin R, He JP, *et al.* Prevalence and correlates of bipolar spectrum disorder in the world mental health survey initiative. Arch Gen Psychiatry. 2011; 68:241-251

Post, RM. Epigenetic basis of sensitization to stres, affective episodes, and stimulants: implications for ilness progression and prevention. Bipolar Disorder. 201; 18:315-324

Merikangas K, Pato M. Recent Developments in Epidemiology of Bipolar in Adults and Children: Magnitude Correlates and Future Directions. Clinical Psychology and Practice. 2009; 16(2): 121-133.

Suppes T, Leverich GS, Keck PE et al. The Stanley Foundation Treatment Outcome Network.II. Demographics and ilness characteristics of the first 261 patients. Journal of Affective Disorders. 2001. 67 (1-3): 45-59

Swann AC, Lafer B, Perugi G, Frye M.A, Bauer M, Bahk VM, Scott J, Ha K, Supples T. Bipolar mixed states: an international society for bipolar disorders task force report of

symptom structure, course of illness and diagnosis. American Journal of Psychiatry. 2013. Jan 17(1): 31-42

CHAPTER 4. VIRGINIA WOOLF– A BRIEF OVERVIEW OF HER FAMILY AND HER GENES

Bennet, M Virginia Woolf and Neuropsychiatry. Netherlands. Springer, 2013.

Bell, Q. Virginia Woolf: a biography. New York. Harcourt Brace Jovanovich; 1972

Dunn. J. Virginia Woolf and Vanessa Bell: A Very Close Conspiracy. London. Hachette Digital; 1990.

Caramagno, TC. The flight of the mind. Berkeley. University of California;1992.

Glendinning, V. Leonard Woolf: a biography. Berkeley: Counterpoint; 2008.T

Jamison, K R. Touched with Fire - Manic Depressive Ilness and the Artistic Temperament.

Pages 224-227. Free Press Paperback. New York, NY, USA. 1994

Koutsantoni K. Manic depression in literature: the case of Virginia Woolf. 2012. Medical Humanities. BMJ Journals. Volume 38, issue 1.

Koutsantoni K and Oakley M. Laura Makeapece Stephen: What was wrong with Woolf's half-sister? English Studies, 2018. DOI: 10.1080/0013838X.2017.1423207

Lee, H. Virginia Woolf.New York: Vintage; 1990.

Maitland, F W and Woolf, V. The Life and Letters of Leslie Stephen. Nabu Press, 2010.

Woolf, V; Schulkind J (Editor). Moments of Being: Autobiographical Writings. Pimlico, London, England, 1990.

Woolf, Virginia; Bell A o (Editor). The Diary of Virginia Woolf: Vol. 1: 1915-1919. Mariner Books, London, England. 1979

Maitland, F W and Woolf, V. The Life and Letters of Leslie Stephen. Nabu Press, 2010.

Stephen, J. D; Gillespie D. F and Steele E (editors). Julia Duckworth Stephen: Stories for Children, Essays for Adults. Syracuse. Syracuse University Press. 1987

CHAPTER 5. VIRGINIA WOOLF AND BIPOLAR DISORDER

Bell, A O (Editor) The Diary of Virginia Woolf, Volume 1. 1915-1919. Mariner Books. London, England. September 1980.

Bell, A O and McNeillie. The Diaries of Virginia Woolf, Volume 3. 1925-1930. Mariner Books. London, England. September 1981.

Bell, Q. Virginia Woolf: a Biography. Harcourt Brace Jovanovich, New York and London. 1972.

Harris, A. Virginia Woolf. Thames and Hudson. London, England, UK. 2011.

Lounsberry, B. Virginia Woolf, The War Without, The War Within: Her Final Diaries and the Diaries She Read. University Press of Florida, Gainesville, Florida, USA. 2018.

Mitchell SW. Fat and Blood: an essay on the treatment of certain forms of neurasthenia and hystheria. Lippincot. London, England. 1884

Panken, S. Virginia Woolf and the lust of creation: a psychoanalytic exploration. The State of New York Press, Albany, NY, USA. 1987

Pearce, JMS. Silas Weir Mitchell and the "rest cure". Journal of Neurology, Neurosurgery and Psychiatry. Volume 75, Issue 3.

Poole, R. The Unknown Virginia Woolf. Cambridge University Press. Cambridge, England. September 1978.

Trombley, S. All that summer she was mad: Virginia Woolf, Female Victim of Male Medicine. Continuum International Publishing Group. 1982.

Trombley. Virginia Woolf and her doctors. PhD thesis. University of Nottingham. http://eprints.nottingham.ac.uk/id/eprint/30895.

Woolf, V. Mrs. Dalloway. Penguim Modern Classics. London, England, UK. May 2000.

Woolf, V; Nicolson N and Trautmann J (Editors). The Letters of Virginia Woolf. Volumes 1-6. Harcourt Brace Jovanovich Publishers. New York and London. 1975.

CHAPTER 6. TRAUMA AND BIPOLAR DISORDER IN VIRGINIA WOOLF

Androutsopoulou, A., Moschona, N., Nioti, N, Rozou., E., Vakondiou, M., Yfanti, K., Zisi, O. (2015). Virginia Woolf's "22 Hyde Park Gate": A narrative analysis of her autobiographical speech on family sexual abuse. Poster presented at the 20th EAP Congress, June 18-21, Athens. Posted at researchgate.net doi:10.13140/RG.2.1.2142.02427.

Daruy-Filho L; Brietzke E; Kluve=Schiavon B; Fabres da S. C; Grassi-Oliveira R. Childhood maltreatment and coping in bipolar disorder. Psychology and Neuroscience. Volume 6, Issue 3. Rio de Janeiro. December 2013.

Daruy-Filho L., Brietzke, E, Lafer, B., & Grassi-Oliveira, R. Childhood maltreatment and clinical outcomes of bipolar disorder. Acta Psychiatrica Scandinava, 124, pp 427-434. 2011

Etain. B., Henry, C., Bellivier, F., Mathieu, F., & Leboyer, M. Beyond Genetics: childhood affective trauma in bipolar disorder. Bipolar Disorder. 10 (8), pp 319-328. 2008

Garno L.J, Goldberg F. J, Ramirez PM and Ritzler BA. Impact of Childhood Abuse on the Clinical Course of Bipolar Disorder. The British Journal of Psychiatry. Volume 186, Issue 2, February 200 5. pp- 121-125

Hyun M, Friedman SD, Dunner DL. Relationshio of childhood physical and sexual abuse to adult bipolar disorder. Bipolar Disorder. 2000 2:131-135,

Konradt E C; Jansen K; Magalhães V. S. P; Pinheiro T. R; Kapczinki P. F; da Silva A. Ricardo; de Souza D. M. L. Early Trauma and mood in youngsters. Archives of Clinical Psychiatry. Volume 40, Issue 3. São Paulo 2013

Laura P. Chen BS, M. Massan Murad MD, Molly L. Paras BS, Kristina M Colbenson BS, Amelia I. Sattler BS, Erin N. Goranson BS, Mohamed B. Elamin MD, Richard J. Seime PhD, Gen Shinozaki MD, Larry J. Prokop MLS, Ali Zirakzadeh MD. Sexual Abuse and Lifetime Diagnosis of Psychiatric Disorders: Systematic Review and Meta-Analysis. Mayo Clinic Proceedings. Volume 85, Issue 7, July 2010. pp 618-629

Lu W, Musser KT, Rosenberg SD, Jankowski MK. Correlates of adverse childhood experiences among adults with severe mood disorders. Psychiatric Services. 2008.59:1018-1026.

Leverich SG and Post RM. Course of bipolar ilness after history of childhood trauma. Lancet. Volume 367, Issue 9516, P1040-1042. April, 2006.

Soledad Romero, Broris Birmaher, David Axelson, Tina Goldstein, Benjamin I. Goldstein, Mary Kay Gill, Ana -Maria Iosif, Martin Keller, Michael A. Strober, Jeffrey Hunt, Christianne Esposito-Smythers, Neal D. Ryan, Henrietta Leonard, Martin Keller. Prevalence and correlates of physical and sexual abuse in children and adolescence with bipolar disorder. Journal of Affective Disorders. Volume 112, Issues 1-3, January 2009. Pages 144-150.

CHAPTER 7. SUICIDE, BIPOLAR DISORDER AND THE CREATIVE PROFESSIONS

Andreasen, N.C (1987). Creativity and Mental Ilness. American Journal of Psychiatry, 144, 1288-1292.

Bron. B, Strack. M & Rudolph, G. (1991). Childhood experiences of loss and suicide attempts. Journal of Affective Disorders, 23, 165-172.

Canetto, S. & Lester, D. (Editors) (1993) Women and suicide. New York: Springer.

Cutter. F. (1983). Art and the wish to die. Chicago. Nelson Hall

Holden, C. (1987). Creativity and the troubled mind. Psychology Today, 21 (4), 9-10.

Jamison, K. R. Night Falls Fast. (2000) First Vintage Books Edition. New York.

Jamison, K. R (1993). Touched with fire. New York. Free Press.

Kaufmann C. J. The Sylvia Plath Effect: Mentall Ilness in Eminent Creative Writers. Journal of Creative Behavior. Volume 35, No 1. First Quarter 2001. https://doi.org/10.1002/j.2162-6057.2001.tb01220.x

Lester, D. (1987) Suicide. In C.S Widom (Ed.) Sex roles and psychopathology, pp 145-156. New York: Plenum.

.Lester, David. Suicide in Creative Women. In Suicide in Creative Women, Chapter 14.

Nova Science Publishers Inc

Menninger, K. (1938). Man against himself. New York: Harcourt Brace & World.

Pretti, A. & Miotto, P. (1999). Suicide among eminent artists. Psychological Reports. 84 (1), pp 291-301.

Stack, S. & Lester, D. (Editors). (2009) Suicide and the Creative Arts.Nova Science Publishers, Inc. New York

CHAPTER 8. CREATIVITY AND BIPOLAR DISORDER

Jamison, K.R (1989). Mood disorders and creativity in British writers and artists. Psychiatry, 52, pp 125-134.

Kaufmann, J.C (2001). Dissecting the golden goose Studying Creative writers. Creativity Research Journal.

Lindauer, M.S (1994). Are creative writers mad? An empirical perspective. In B. M Rieger (Ed.), Dionysius in Literature: Essays on literary madness. Bowling Green, Ohio: Bowling Green State University Popular Press.

Ludwig, A.M (1995) The price of greatness. New York: Guilford Press.

Ludwig A.M (1998) Method and madness in the arts and sciences. Creativity Research Journal, 11 (2), 93-101.

Pirto, J (1998). Themes in the lives of successful contemporary U.S women creative writers. Roeper Review, 21 (1), 60-70

Pohlmann, L. (1996). Creativity, gender and the family: A study of creative writers. Journal of Creative Behavior, 30 (1), pp 1-24.

Post, F. (1996). Creativity and Psychopatology: A study of 291 world-famous men. British Journal of Psychiatry, 165 (1), pp 22-34.

Post, F. (1996). Verbal creativity, depression and alcoholism: An Investigation of one hundred American and British writers. British Journal of Psychiatry, 168 (5), pp 545-555.

Rothemberg, A. (1990) Creativity and madness. Baltimore: Johns Hopkins University Press.

Simonton, D.K. (1994). Greatness. New York: Guilford

Waddell, C. (1998). Creativity and mental ilness. Is there a link? Canadian Journal of Psychiatry, 43 (2), 166-172.

CHAPTER 10. VIRGINIA WOOLF'S FINAL DAYS AND SUICIDE

Bell, Vanessa. Selected Letters of Vanessa Bell. Ed Regina Marler. Wakefield Rhoda. Island & London: Moyer Bell. 1998.

Congenial Spirits: The Selected Letters of Virginia Woolf. Ed. Joanne

Trautmann Banks. San Diego, New York, London: Harcourt Brace Jovanovich, Publishers.

The Essays of Virginia Woolf, Volume V: 129 -1932. 2009. Ed. Stuart N Clarkke. 1st US edition. Boston, New York: Houghton Mifflin Harcourt, 2010.

The Diary of Virginia Woolf. Volume Three: 1925-1930. 1980. Ed Anne Oliver Bell and Matthew McNeillie. 1st Harvest edition. San Diego, New York, London: A Harvest Book, Harcourt Brace & Company. 1981.

The Diary of Virginia Woolf. Volume Four: 1931-1935. 1980. Ed Anne Oliver Bell and Matthew McNeillie. 1st Harvest edition. San Diego, New York, London: A Harvest Book, Harcourt Brace & Company. 1983.

The Diary of Virginia Woolf. Volume Four: 1936-1941. 1980. Ed Anne Oliver Bell and Matthew McNeillie. 1st Harvest edition. San Diego, New York, London: A Harvest Book, Harcourt Brace & Company. 1985.

The Letters of Virginia Woolf, Volume III: 1929-1931. 1978. Ed. Nigel Nicholson and Joane Trautmann. 1st American Edition. New York and London: Harcourt Brace Jovanovich.

The Letters of Virginia Woolf. Volume IV: 1936-1941. 1980. Ed. Nigel Nicholson and Joanne Trautmann. 1st American Edition. New York and London: Harcourt Brace ovanovich, 1980.

Woolf, Leonard. The Letters of Leonard Woolf. Ed Frederic Spotts. San Diego, New York, London: Harcourt Brace Jovanovich, 1989.

Woolf, Leonard. The Autobiography of Leonard Woolf, volumes 3 to 5:

Beggining Again: An Autobiography of the Years 1911 to 1918. New York and London: A Harvest Book, Harcourt Brace Jovanovich, 1963.

Downhill all the Way: An Autobiography of the Years 1199 to 1939. New York and London: A Harvest Book, Harcourt Brace Jovanovich, 1967.

The Journey Not the Arrival Matters: An Autobiography of the Years 1939 to 1969. New York and London: A Harcourt Book. Harcourt Brace Jovanovich. 1969.

Woolf, Virginia. Between the Acts (1941). 1st Harvest edition. San Diego, New York, London: A Harvest Book, Harcourt, Inc, 1970.

Woolf, Virginia. Orlando: A Biography (1928). San Diego, New York, London: A Harvest Book, Harcourt, Inc., 1956.

Woolf, Virginia. The Waves (1931) Storck Press. 2016

Woolf, Virginia. Moments of Being. 1976. Ed. Jeanne Schulkind. 2a edition. San Diego, New York, London: A Harvest Book, Harcourt Brace & Company, 1985:p. 61-159